ARTICLE:THE FIVE HABITS: BUILDING CROSS-CULTURAL COMPETENCE IN LAWYERS

Fall, 2001

Reporter 8 Clinical L. Rev. 33

Length: 29857 words

Author: Susan Bryant*

* Associate Professor, City University of New York School of Law (CUNY). As described in the epilogue, this article grows out of a collaborative project with Jean Koh Peters, to whom I am deeply grateful. In our work on developing the Habits and the teaching module to teach the Habits, Jean and I were aided by many wonderful colleagues, students, and staff. I want to thank those colleagues who first worked with me on issues of diversity in AALS presentations, including Victor Goode, Isabelle Gunning, Steve Hardwell, and Jennifer Rockow. Jean and I have wonderful colleagues who have taught cross-cultural lawyering using these materials or watched us use them and given us feedback on the Habits and these materials. They include my colleagues at CUNY: Beryl Blaustone, Rhonda Copelon, Sam Dulberg, Ellen Fried, Gail Gray, Pamela Goldberg, Sharon Hom, Ron Lindeman, Steve Loffredo, Joe Rosenberg, and Susan Taylor; and Jean's colleagues at Yale: Carroll Lucht, Michael Pinard, Jay Pottenger, and Steve Wizner. Jean's colleague and our friend, Kathleen Sullivan, passed away before this article was published. She was an enthusiastic supporter and contributor to our work. She is missed deeply. I am especially grateful to Maria Arias, my co-teacher for ten years at CUNY, for her insights and conversations about teaching in general and about cross-cultural issues, and to my colleagues, Bob Seibel and Alice Morey, and to Harvey Weinig for their careful reading and suggestions. I also thank the following students from Yale and CUNY whose research, thoughtful feedback, and support made this work possible: Sonia Cardia, D. Scott Carlton, Melissa Chiang, Alexandra Joncas, Judith Karpatkin, Cary Berkeley Kaye, Pavita Krishnaswamy, Leondra Kruger, Judy Lee, Nicola Mrazek, Sharmila Rampersaud, and Anjan Sahni. Two friends of Jean's, Karen Johnson and Jim Lynch, listened and commented on the habits at two critical early moments. I thank the staff of the Yale's Jerome N. Frank Legal Services Organization, in particular Kathleen Driscoll, Maureen Furtak, and Linda McMahon, and the staff of CUNY's clinical program, in particular Bernice Cohn, Lillian Gonzalez, Verleatha Hill, and Dorothy Matthew. The Open Society Institute, Emma Lazarus Fund, provided support for the conference, our work and the publication of the student materials on which this article is based. I thank OSI for recognizing that law schools have a role to play in serving immigrant communities, Dean Kristin Glen of CUNY for her creative energy in proposing the idea to OSI and bringing it to fruition at the CUNY School of Law, and Alizabeth Newman, Director of CUNY's Immigrant's Initiatives, for her work in organizing and publishing the teaching modules. Jean's work was also funded through the generosity of Dean Anthony Kronman and the Yale Law School. Finally, I thank my husband Larry and my children, Alison and Zach, and Jean's children, Lizzie and Chris, who made trips to New Haven even more fun.

LexisNexis Summary

... By describing classroom discussions and exercises used to teach the Habits, the article illustrates methods that clinical teachers can use to have more inclusive classroom conversations while building cross-cultural skills. ... Habit Two asks the student first to identify the differences and similarities between the client and the legal system and between the lawyer and the legal system. ... If helpful, students can use diagrams to chart the differences and similarities as they did in Habit One to visualize different dyads - lawyer/client, client/legal system, lawyer/legal system. ... Habit Four encourages conscious attention to the process of communication - a skill and perspective that clinical teachers have used to improve interviewing skills in all attorney-client interactions. ... Although Habit Four is a habit that can be done in the moment, beginning lawyers often have difficulty paying attention to both the process and content of an interview at the same time. ... Finally, if the client has come from another country, Habit

Four encourages students to gather information about expectations the client may have about the lawyer and the legal system by learning about how the legal system works in the client's country of origin. ... Habit Two develops skills of analyzing the effects of similarities and differences among the client, legal system and the lawyer. ...

Text

[*33]

"Yet the challenge confronts us: Build a unified society without uniformity." ¹

This article describes a process called "the Habits" that was developed by Professors Bryant and Jean Koh Peters that can be used by lawyers to increase their cross-cultural competence. By outlining [*34] and giving examples of the role that culture plays in decision making, communication, problem solving, and rapport building, the article demonstrates the importance of lawyers learning cross-cultural concepts and skills. The article shows how developing the Five Habits increases cross-cultural competence. By describing classroom discussions and exercises used to teach the Habits, the article illustrates methods that clinical teachers can use to have more inclusive classroom conversations while building cross-cultural skills. In the epilogue to the article, Professors Bryant and Peters describe the collaborative process they used to develop the Habits.

Introduction

A lawyer and translator approach the reception area to greet a client and the following dialogue occurs:

Reception Area

[SEE DIALOGUE IN ORIGINAL]

How does the lawyer proceed with the interview at this point? Does she bring the brother into the interview? Does she explain to the brother that, for confidentiality reasons, she must interview the sister alone? Does she view the brother as connected to the dispute with the landlord? What assumptions does the lawyer have about why the brother is there and what the potential benefits and costs are for including or excluding the brother? To what extent are the lawyer's assumptions and responses to the situation based on the lawyer's experiences [*35] and culture? How can lawyers correctly attribute meaning to their clients' behavior and thereby make better lawyering decisions?

This article explores ways in which lawyers might answer these questions by focusing on the role culture plays in lawyering. The article presents a process, referred to as "the Habits," which lawyers can use to avoid cultural blinders and recover from cultural blunders when they occur. It also explores ideas that clinical teachers can use in teaching about issues of difference, identity, bias and stereotyping. ² The Habits, which Jean Koh Peters and I

¹ R.W. Terry, Authenticity: Unity Without Uniformity, in The Promise of Diversity: Over 40 Voices Discuss the Strategies for Eliminating Discrimination in Organizations 113-14 (E.Y. Cross, J.H. Katz, F. A. Miller, & E. W. Seashore eds., 1994).

² The Habits are described in Section Three of this article. Materials designed to be used in teaching the Habits were published by the CUNY Immigrant's Initiatives Project, Integrating Immigrant's Perspectives Across the Curriculum, and can be downloaded from <<u>http://clinic.law.cuny.edu/clea/multiculture></u>. These materials are part of a larger project which is designed to integrate immigrants' perspectives throughout the curriculum and which includes modules in seven other subjects. As a later section of the article will explain, Jean and I encourage you to use the teaching materials in any way that seems useful. I have sometimes used articles to develop outlines for students. I hope you will feel free to edit these materials and make them useful for your students. Although this article's examples focus on the contexts of civil and administrative litigation, teachers in other areas can use the materials by including or substituting their own stories in the materials. In changing the materials, Jean and I ask that you note that "The materials being distributed are based on materials written by Susan Bryant and Jean Koh Peters. They have been edited (and /or added to) by x." If you would like to put your changed materials on the web site for others to use, please contact me and I will arrange to do that. If you develop an interesting case study or simulation or video for use, please developed in a process described in the epilogue, are still a "work in progress." ³ We continue to study the efficacy of the Habits for our students as they explore their own development of cross-cultural skills.

Many clinical teachers have recognized the importance of teaching diversity issues in the clinic. ⁴ A number of presentations at AALS [*36] Clinical Teachers Conferences and entire conferences sponsored by AALS and by SALT have been devoted to exploring ways in which diversity can be taught in the clinic and classroom. ⁵ This article focuses on teaching diversity through use of a framework of cross-cultural lawyering. In deciding how to teach cross-cultural lawyering, clinical teachers need to identify how to integrate it into the overall goals of their clinics and to set specific goals for student competence in this area.

Each clinic may answer this inquiry slightly differently, depending on the faculty, students and clients, and on the level of change that the clinic wants to accomplish. On the micro level, a clinic may teach cross-cultural lawyering to improve representation of clients and to introduce students to cross-cultural theory that they will be able to apply to practice in an ever increasingly diverse legal profession. On the macro level, a clinic may teach cross-cultural perspectives and skills to enable students to help build a more just legal system. ⁶ [*37] Often, the priority given

send it along and we can post it. New teaching ideas can also be placed on the site. In that way, we do not have to wait for a new law review article to spread ideas about teaching about issues of difference and similarity.

³ We recognize that, as Margaret Montoya noted, "engaging in public dialogue about difference ... requires practice because we so often can, and do get things wrong." We constantly revise vocabularies and theories and employ "provisional pedagogical and interactive strategies." Margaret Montoya, Voicing Difference, <u>4 Clin. L. Rev. 147, 152 (1997).</u>

4 See Jane Harris Aiken, Striving to Teach Justice, Fairness and Morality, 4 Clin. L. Rev. 1 (1997); David Dominguez, Beyond Zero-Sum Games: Multiculturalism As Enriched Law Training For All Students, 44 J. Legal Educ. 175 (1994); Leslie G. Espinoza, Legal Narratives, Therapeutic Narratives: The Invisibility and Omnipresence of Race and Gender, 95 Mich. L. Rev. 901 (1997); Melissa Harrison & Margaret E. Montoya, Voices/Voce in the Borderlands: A Colloquy on Re/Constructing Identities in Re/Constructed Legal Spaces, 6 Colum. J. Gender & Law 387 (1996); Bill Ong Hing, Raising Personal Identification Issues of Class, Race, Ethnicity, Gender, Sexual Orientation, Physical Disability, and Age in Lawyering Classes, 45 Stan. L. Rev. 1807, 1810 (1993); Michelle Jacobs, People From the Footnotes: The Missing Element in Client-Centered Counseling, 27 Golden Gate U. L. Rev. 345, 372 (1997); Kimberly O'Leary, Using Difference Analysis to Teach Problem-Solving, 4 Clin. L. Rev. 65, 72 (1997); Suellyn Scarnecchia, End Gender and Race Bias Against Lawyers: A Classroom Response, 23 U. Mich. J. L. Reform 319 (1990), Christine Zuni Cruz, [On the] Road Back In: Community Lawyering In Indigenous Communities, 5 Clin. L. Rev. 557 (1999). See also Robert F. Cochran, Jr., John M.A. DiPippa & Martha M. Peters, The Counselor-at-law: A Collaborative Approach to Client Interviewing and Counseling, Teacher's Manual 163 (1999). New works that advocate this kind of teaching include: Marjorie A. Silver, Emotional Competence, Multicultural Lawyering and Race (paper presented at the Second International Conference on Therapeutic Jurisprudence, Cincinnati, Ohio, May 4, 2001); and Shin Imai, Thoughts on Community, Critical Race Praxis, and Clinical Pedagogy or, "What I Teach in My Students" (paper presented at the New York Law School Clinical Theory Workshop, April 27, 2001).

⁵ Presentations at AALS conferences, workshops and annual meetings have become so frequent and valuable that I hesitate to name one for fear of omitting some. Over the years, as our teaching has gotten more sophisticated on these topics, the presentations have gotten better and challenged our thinking. The AALS Clinical Workshop in Washington in 1991 was the first clinical workshop devoted to these issues and it was a good beginning. The recent 2000 AALS Clinical Conference in Albuquerque on culture and the AALS Clinical Section Annual Meeting program on diversity issues in hiring stand out for me as recent, wonderful examples of our progress. See Jon Dubin, Faculty Diversity as a Clinical Legal Education Imperative, <u>51</u> <u>Hastings L.J. 445 (2000)</u>, for a description of the annual program. SALT teaching conferences have also been useful in presenting examples of teaching about diversity and building a more just society through structuring classroom conversations in traditional courses and devising action projects for those courses. By sharing teaching strategies and presenting conceptual frameworks, colleagues in SALT and the clinical community empower those who care about these issues by teaching us how to teach and reflect about culture, discrimination, bias and justice.

⁶ Whatever the specific goals, adult learning theory tells us that perspective transformation can occur "from efforts to understand a different culture with customs that contradict our previously accepted suppositions." Fran Quigley, Seizing the Disorienting Moment: Adult Learning Theory and the Teaching of Social Justice in Law School Clinics, <u>2 Clin. L. Rev. 37, 72 n.55</u> (<u>1995</u>), citing Jack Mezirow, Transformative Dimensions Of Adult Learning 211 (1991). In conducting a needs assessment prior to designing training, cross-cultural trainers ask, for example, whether the goal is to develop skills for employees or other to these micro and macro objectives will influence teachers to structure the learning to give greater weight to one goal over the other. ⁷

I hope this article helps clarify some of the choices that we as teachers make in teaching about culture. Often, in planning classes, we think about the topics we want to teach and put together materials without a clear articulation of specific learning goals or an explicit conversation with ourselves about the implicit pedagogical assumptions regarding how our students learn. In developing the Habits, articulating materials that described them, and designing different ways to teach them, Jean and I kept questioning ourselves about two core issues: (1) what is effective cross-cultural lawyering and (2) how can we help ourselves and our students learn to be effective cross-cultural lawyers? This article attempts to answer these questions and to describe the process we went through in developing our answers.

Section One identifies the significance of culture to the lawyering process. In this section, I hope to answer the critics who think that good lawyering is "culture-neutral" and that therefore any attention to the study of cross-cultural skills and perspectives is unnecessary or not an important enough aspect of lawyering to be highlighted in our teaching. Section Two describes our theory of cross-cultural lawyering competence and identifies choices that teachers can make in setting competency goals for their students. Section Two also lays out the pedagogical theory that underlies the Habits and the resistance that we can expect from both students and ourselves. Section Three summarizes the Habits⁸ and Section Four identifies ways to teach the Habits. Through application of the Habits, students will learn the perspectives and skills described in Section Two so as to become effective cross-cultural lawyers.

[*38]

I. Teaching Skills Of Cross-Cultural Lawyering As An Important Part Of Teaching Good Lawyering

For many years, companies choosing employees for overseas work and schools selecting students for overseas study gave personality tests to explore who would make good travelers, adjust more readily to cross-cultural encounters and respond appropriately in culturally sensitive ways. Generally, people who were flexible, less judgmental and more reflective were viewed as having the right personality to work in cross-cultural environments. ⁹ This notion that success is dependent on personality has been replaced by the idea that cross-cultural competence is a skill that can be taught. ¹⁰ As with learning of most skills, there are those who seem to have some natural talent and others who, as a result of experience, have developed special insights into cross-cultural work. However, everyone has the capacity to become more proficient at cross-cultural interaction and communication skills.

trainees or whether changing the organizational culture and systems or the community and society are goals for the training. The overall goals affect the amount of time allocated to the training, the curriculum and the measurement of success.

⁷ For example, Jane Aiken's goal of teaching justice presents a focus on teaching one core difference that is a framework for others: privilege. Aiken, supra note 4. This lens and goal are slightly different in emphasis than a goal of teaching students to use a difference analysis for their lawyering, see O'Leary, supra note 4, or teaching empathy, see Stephen Ellmann, Empathy and Approval, <u>43 Hastings L.J. 991 (1992)</u>. Like those described in this article, the methods outlined in these articles, and the others in note 4 supra, teach students important skills for good cross-cultural lawyering and for social change. How the students use these skills and insights within the lawyering context are part of the conversations that teachers and students can have once they are aware of the insights that all of these methods raise for students.

⁸ A full set of materials and accompanying worksheets is available. See supra note 2. For those who teach in family or child protective clinics, Jean has a chapter on the Habits in her book. See Jean Koh Peters, Representing Children In Child Protective Proceedings: Ethical and Practical Dimension (Supp. 2000).

⁹ See Improving Intercultural Interactions: Modules for Cross-Cultural Training Programs (Kenneth Cusher & Richard W. Brislin eds., 1997) (hereinafter "Improving Intercultural Interactions").

¹⁰ Koh Peters, supra note 8.

Almost all professions and businesses now recognize the importance of building cross-cultural skills. ¹¹ The United States is increasingly a multi-cultural country with a greater understanding that the "melting pot" did not happen. ¹² Materials have been developed and [*39] courses offered for training teachers, doctors, social workers, psychologists and psychiatrists about cross-cultural issues in their professions. ¹³ As our world becomes more interactive, lawyers and clients inevitably will interact with those who are culturally different. Those whom we assume to be just like us may turn out not to be in some important ways while those whom we assume to be different may, in fact, not be so different. ¹⁴

¹¹ For example, businesses recognize that, with improved skills, they will be more competitive, increase retention, improve performance, and be better prepared to work in a multi-cultural business world. The American Medical Association has developed a broad-based initiative to establish cultural competence as the "Fifth Physician Competence," and has produced a collection of resources for physicians to advance this effort called The Cultural Competence Compendium (1999), available at http://www.ama-assn.org/ama/pub/category/2661.html. See also Assuring Cultural Competence in Health Care: Recommendations for National Standards and Outcomes-Focused Research Agenda, at http://www.omhrc.gov/clas/ds.htm; Diversity RX, at http://www.omhrc.gov/clas/ds.htm; Diversity RX, at http://www.diversityrx.org/gucdc.georgetown.edu/nccc/cultural.html, home page for the National Center for Cultural Competence, which was designed to improve medical care through linguistic and culturally competent services. Some legal organizations have begun to pay attention to these issues as well. This includes, e.g., the Legal Language Access Project (LLAP), a project that grew out of the joint work of the East Bay Asian Consortium and the Alameda County Bar Association Volunteer Legal Services to Culturally Diverse Communities," to increase awareness and change attitudes and practices

12 Regardless of the parameters chosen, data from the 2000 Census of the U.S. and associated studies and analyses reveal increasing diversity in all fifty states. From more conventional indices, such as race and ethnicity, to languages spoken in households and percentage increases in unmarried partner households or same-sex households, the last decade has witnessed a significant change in the demographic makeup of the population of the United States. For example, more than 10.5 million people reported in a national survey conducted in conjunction with the 2000 Census that they primarily spoke a language other than English, up from 6.5 million in 1990. See D'Vera Cohn & Sarah Cohen, Census Sees Vast Change in Language, Employment, Washington Post, Aug. 6, 2001, at A01. So also, a number of gay and lesbian organizations across the nation have compiled statistics based on the number of unmarried partner or same-sex partner households reported in census data, and their unanimous conclusion has been that the number of such households has increased dramatically. See, e.g., the websites of the National Gay and Lesbian Task Force and the Human Rights Campaign at http://www.ngltf.org and <http://www.hrc.org>, respectively. Data from the Census Bureau (available at <http://www.census.gov/population/cen2000/ phc-t1/tab04.pdf>) itself reveals increasing racial and ethnic diversity, with a 21.5% reported difference in Black or African-American respondents between 1990-2000, and a 57.95 difference in Latino or Hispanic respondents in the same time period. The conclusion is irrefutable: Americans are more likely to be racially, linguistically, and sexually diverse than they were a decade ago.

¹³ A wealth of literature exists in the field of cross-cultural training. I encourage those of you who want to deepen your understanding of culture, cross-cultural competence and how to teach it to read this literature. Sage Publications in Thousands Oaks, California has published a number of books on cross-cultural training. Some recommendations include: Handbook of Intercultural Training (Dan Landis & Rabi S. Bhagat eds., 2d ed. 1996); Intercultural Interactions: A Practical Guide (Kenneth Cusher & Richard W. Brislin eds., 1996) (hereinafter "Intercultural Interactions"); and Improving Intercultural Interactions, supra note 9.

¹⁴ We might have thought about this change as us or them assimilating and losing the sameness or difference. Now we see that it is more complicated and recognize that people can cross borders and at times be different and at other times remain the same. Margaret Montoya has used the idea of border crossing to explore the contingent nature of outsider and insiderness, Montoya, supra note 3. By applying the concept of borderlands used by Gloria Anzaldua, D. Emily Hicks, and Peter Laren, Professors Montoya and Harrison teach a way to develop empathy and connection through identities that are both self and the Other. These identities borrow from the experience of border crossers who live in "bilingual, bicultural, biconceptual reality." Harrison & Montoya, supra note 4, at 398. As our profession becomes increasingly diverse, ¹⁵ the tensions created by difference offer great potential for creative change. ¹⁶ These same tensions, however, could result in negative judgments and **[*40]** misunderstanding. By teaching students how to recognize the influence of culture in their work and to understand, if not accept, the viewpoint of others, we provide students with skills that are necessary to communicate and work positively with future clients and colleagues.

To become good cross-cultural lawyers, students must first become aware of the significance of culture on themselves. Culture is like the air we breathe - it is largely invisible and yet we are dependent on it for our very being. Culture is the logic by which we give order to the world. ¹⁷ Culture gives us our values, attitudes and norms of behavior. We are constantly attaching culturally-based meaning to what we see and hear, often without being aware that we are doing so. ¹⁸ Through our invisible cultural lens, we judge people to be truthful, rude, intelligent or superstitious based on the attributions we make about the meaning of their behavior.

By teaching students cross-cultural lawyering skills and perspectives, we make the invisible more visible and thus help students understand the reactions that they and the legal system may have towards clients and that clients may have towards them. When two people (such as two student co-counsels or a student and a clinical teacher) working on the same case differ, we have the opportunity to explore why we are giving different meaning to the same behavior and words. By teaching the students about the influence of culture on their practice of law, we give them a framework for analyzing the changes that have resulted in their thinking and values as a result of their legal education. The law, as well as the legal system within which it operates, is a culture with strong professional norms that gives meaning to and reinforces behavior. How legal education influences the choices that students do or do not see is an important part of the cross-cultural analysis.¹⁹

Cross-cultural lawyering occurs when lawyers and clients have different ethnic or cultural heritages and when they are socialized by [*41] different subsets within ethnic groups. By this definition, everyone is multi-cultural to some degree. ²⁰ Cultural groups and cultural norms can be based on ethnicity, race, gender, nationality, age, economic status, social status, language, sexual orientation, physical characteristics, marital status, role in family, birth order, immigration status, religion, accent, skin color or a variety of other characteristics. ²¹

¹⁵ See <<u>http://www.abanet.org/legaled/statistics/></u>, showing increasing numbers of wo-

men and minority students enrolled in J.D. programs between 1984-2000. For example, the number of minority students enrolled in full-time J.D. programs more than doubled in the period between 1984 and 2000, going from 11,917 to 25,753, and the number of women in law school went up from 46,897 to 60,633 in the same time period.

¹⁶ In an earlier article, I discussed the benefits of joint work that lawyers with diverse perspectives can bring to their work. In the article, I argued that joint work could take advantage of the increasing diversity of the bar. Lawyers who plan for and appreciate difference in work style, communication styles and processing values will be able to collaborate more effectively. See Susan Bryant, Collaboration in Law Practice: A Satisfying and Productive Process for a Diverse Profession, 17 Vt. L. Rev. 459 (1993).

¹⁷ Raymonde Carroll, Cultural Misunderstandings: The French-American Experience 2 (1988). Objective culture includes that which we observe, including artifacts, food, clothing, names. It is relatively easy to analyze and identify its use. Subjective culture refers to the invisible, less tangible aspects of behavior. People's values, attitudes, beliefs are kept in people's minds. Most cross-cultural misunderstandings occur at the subjective culture level. Intercultural Interactions, supra note 13, at 6.

¹⁸ Ethnocentrism occurs when a person uses his or her own value system and experiences as the only reference point from which to interpret and judge behavior. Because socialization is such a powerful process, we are hardly aware that different realities can exist and we therefore make ethnocentric judgments about others all the time, believing the judgment is right rather than just a viewpoint. Intercultural Interactions, supra note 13, at 5.

- ¹⁹ See infra notes 174-79 and accompanying text.
- ²⁰ Intercultural Interactions, supra note 13, at 10.
- ²¹ See Habit One, discussed infra notes 121-32 and accompanying text.

In teaching about the importance of culture to lawyering, we want to avoid reinforcing stereotypes. By using a broad definition of culture, we hope to teach students that no single characteristic will completely define the lawyer's or client's culture. ²² For example, if we think about birth order alone as a cultural characteristic, we may not see any significance to this factor. Yet, if the client (or lawyer) comes from a society where "oldest son" has special meaning in terms of responsibility and privilege, identification of the ethnicity, gender or birth order alone will not be enough to alert the lawyer to the set of norms and expectations for how the "oldest son" is expected to behave. Instead, the lawyer needs to appreciate the significance of the combination of ethnicity, birth order, and gender to fully understand this aspect of the client's culture. A woman from the same culture may understand these responsibilities and privileges better than an outsider and yet, because her experiences are different, still may not fully understand.

A broad definition of culture recognizes that no two people can have exactly the same experiences and thus no two people will interpret or predict in precisely the same ways. Culture is enough of an abstraction that people can be part of the same culture, yet make different decisions in the particular.²³ People can also reject norms and values from their culture. As we recognize these individual differences, we also know that sharing a common cultural heritage with a client tends to improve our predictions and interpretations and to reduce the likelihood of misunderstandings.

When lawyers and clients come from different cultures, several aspects of the attorney-client interaction may be implicated. The capacity [*42] to form trusting relationships, to evaluate credibility, to develop client-centered case strategies and solutions, to gather information and to attribute the intended meaning from behavior and expressions are all affected by cultural experiences. By using the framework of cross-cultural interaction, students can learn how to anticipate and name some of the difficulties they or their clients may be experiencing. By asking students as part of the cross-cultural analysis to identify ways in which they are similar to clients, we identify the strengths of connection. We also alert students who see themselves as "the same" as the client to be mindful of differences so that they do not substitute their own judgment for the client's as a result of over-identification or transference.²⁴

Lawyers and clients who do not share the same culture face special challenges in developing a trusting relationship in which genuine and accurate communication can occur. ²⁵ By teaching students concepts like "insider" and "outsider" status, we educate students as to why some lawyers and clients may experience great difficulties in building a relationship in which advice is accepted and information is exchanged freely. When the client's culture fosters a significant distrust of outsiders ²⁶ or of the lawyer's particular culture, the lawyer must work

²² Critical race feminist theorists have pointed out the importance of intersectionality in recognizing, for example, that women of color have different issues than white women or men of color. The intersectionality of race and gender gives women of color a different vantage point and life experience. See Kimberle Crenshaw, Mapping the Margins: Intersectionality, Identity Politics, and Violence Against Women of Color, <u>43 Stan. L. Rev. 1241, 1249 n.29 (1991)</u>; Angela P. Harris, Race and Essentialism in Feminist Legal Theory, <u>42 Stan. L. Rev. 581 (1990)</u>; see also Harrison & Montoya, supra note 4. Professors Montoya and Harrison discuss the importance of seeing multiple and changing identities.

²³ Carroll, supra note 17, at 3. See also David B. Wilkins, Essay: Identities and Roles: Race, Recognition, and Professional Responsibility, <u>57 Md. L. Rev. 1502, 1533 (1998).</u>

²⁴ See discussion of difference and sameness, infra notes 121-32 and accompanying text.

²⁵ Hing, supra note 4, at 1809 (describing impact on rapport of 27-year-old male Chinese American Lawyer representing a 30-year-old African-American and pointing out the value of developing sensitivity to personal identity issues for building rapport with clients).

²⁶ The insider/outsider group distinction is one of the core themes in cross-cultural interactions. Handbook of Intercultural Training, supra note 13, at 189. Historical struggles between native countries of the lawyer and client or situations in which the lawyer's or client's native countries have dominated the other's country can create difficult power dynamics between lawyer and client. Racial discrimination, both historical and current, by Anglo-Americans against African-Americans can have significant influences on the lawyer-client relationship.

especially hard to earn trust in a culturally sensitive way. By teaching the concept of "insider" and "outsider" status, before students form a view of clients as "holding back," "lying," or "being unhelpful," we allow students to have a more nuanced hypothesis about what is occurring in these relationships.²⁷

Even in situations in which trust is established, students may experience cultural differences that significantly interfere with lawyers' and clients' capacities to understand one another's goals, behaviors and communications. Cultural differences often cause us to attribute different meaning to the same set of facts. One important goal of [*43] cross-cultural training is to help students make isomorphic attributions, i.e., to attribute to behavior and communication that which is intended by the actor or speaker. ²⁸ Students who are taught about the potential for misattribution can develop strategies for checking themselves and their interpretations. ²⁹

Inaccurate attributions can cause lawyers to make significant errors in their representation of clients. Imagine a lawyer saying to a client, "If there is anything that you do not understand, please just ask me to explain" or "If I am not being clear, please just ask me any questions." The lawyer might assume that a client who does not then ask for clarification surely understands what the lawyer is saying. However, many cultural differences may explain a client's reluctance to either blame the lawyer for poor communication (the second question) or blame himself or herself for lack of understanding (the first question). ³⁰ Indeed, clients from some cultures might find one or the other of these results to be rude and, therefore, will feel reluctant to ask for clarification for fear of offending the lawyer or embarrassing himself. ³¹

Cultural differences may also cause lawyers and clients to misperceive body language and judge each other incorrectly. ³² For an everyday example, take nodding while someone is speaking. In some cultures, this gesture indicates agreement with the speaker; in others, however, it simply indicates that the listener is hearing the speaker. Another common example involves eye contact. In some cultures, looking someone straight in the eye is a statement of open and honest communication while a diversion of eyes signals dishonesty. In other cultures, however, a diversion of eyes is a sign of respect. Students need to recognize these differences and plan for a representation strategy that takes them into account.

More generally, students need to be taught that concepts of credibility are very culturally determined. In examining the credibility of a story, lawyers and judges often ask whether the story makes "sense" [*44] as if "sense" were neutral. ³³ Consider, for example, a client who explains that the reason that she left her native country was that God

²⁸ Dharm P.S. Bhawuk & Harry C. Triandis, The Role of Culture Theory in The Study of Culture and Intercultural Training, in Handbook of Intercultural Training, supra note 13, at 17.

²⁹ See Habits One, Two, and Three, infra notes 121-39 and accompanying text.

³⁰ People in cultures that are self-effacing tend to blame problems on themselves and not the other whereas those in self-aggrandizing cultures tend to blame problems on others. An individual from a face-saving, collectivist culture would hesitate to challenge the lawyer so as to cause them to lose face. See Richard Brislin & Tomoko Yoshida, Intercultural Communication Training: An Introduction 95 (1994).

³¹ What's a lawyer to do if she cannot just tell the client to "feel free to ask me questions"? Habit Four, infra notes 140-54 and accompanying text, identifies ways that lawyers can "test" comprehension by clients.

³² See Brislin & Yoshida, supra note 30, at 91.

³³ As Professor Hing recognized in one of the first articles written about the influence on lawyering of what he called "personal identification issues," "common sense, without training, is dangerously fashioned by our own class, race, ethnicity/culture, gender and sexual background." Hing, supra note 4, at 1810. See also Carolyn Grose, A Field Trip to Benetton and Beyond: Some Thoughts On 'Outsider Narrative' in the Law School Clinic, <u>4 Clin. L. Rev. 109</u>, <u>116 (1997)</u>. "The supposedly objective

²⁷ Professor Jacobs identifies how an African-American client can be erroneously labeled "difficult" by a white lawyer who fails to understand the significance of racial differences. The lawyer may be sending signals to the client that reinforce racial stereotypes, may be interpreting behavior incorrectly, and, therefore, may be unconsciously failing to provide full advocacy. Jacobs, supra note 4, at 345.

appeared to her in a dream and told her it was time to leave. If the time of departure is critical to the credibility of her story, how will the fact-finder evaluate the client's credibility? Does the fact-finder come from a culture where dreams are valued, ³⁴ where an interventionist God is expected, or where major life decisions would be based on these expectations or values? Will the fact-finder, as a result of differences, find the story incredible or indicative of a disturbed thought process or, alternatively, as a result of similarities, find the client credible? ³⁵

Categorization differences may cause lawyers and clients to view different information as relevant. ³⁶ Students who describe clients as "wandering all over the place" may be working with clients who categorize information differently than the students or the legal system. Lawyers and clients who have different time and space orientations may have difficulty understanding and believing each other. ³⁷ If a [*45] lawyer whose culture is oriented to hour, day, month, and year tries to get a time-line from a client whose culture is not oriented that way, she may incorrectly interpret the client's failure to provide the information as uncooperative, lacking intelligence, or, worse, lying. ³⁸ Clients who are unable to tell a linear time-related story may also experience the same reaction from judges and juries if the client's culture is unknown to the fact finders.

In other settings, the distinction between individual and collective cultures has been called the most important concept to grasp in cross-cultural encounters. ³⁹ Understanding this distinction and the differences that flow from it are also critically important for lawyers to understand. Teaching students to recognize some of the differences between individual and collective cultures will help them see how clients and lawyers define problems, identify

standard of truth on which the criticisms rely is in fact just one <elip> point of view.... Insiders refuse to hear and/or believe outsiders' stories because they often conflict with the insider's understanding of the outsider." <u>Id at 118.</u> See also Naomi R. Cahn, Inconsistent Stories, <u>81 Geo. L.J. 2475, 2515 (1993)</u>, as quoted in Harrison & Montoya, supra note 4, at 426. I am not suggesting that the student refrain from assessing whether the story would make sense to the fact-finder. Obviously, an important part of case preparation is identifying differences between the fact-finder and client that may inhibit a fact-finder's capacity to understand the client and her worldview. Habit Two is designed to get the student to think through those issues. See infra notes 133-136 and accompanying text.

³⁴ Different cultures view rituals and superstitions differently. Rituals and Superstitions are among the eighteen categories of knowledge about cultures that cross-cultural trainers have identified. See Handbook of Intercultural Training, supra note 13, at 189. One group's ritual is viewed by another as superstition.

³⁵ My colleague Sharon Hom shared this vignette with me as well as her students' different reactions to it. See also Marc A. Fajer, Authority, Credibility, and Pre-Understanding: A Defense of Outsider Narratives in Legal Scholarship, <u>82 Geo. L.J. 1845</u>, <u>1856 (1994)</u> ("Faced with a conflict between deep-seated beliefs and a contradicting story, some people may adjust their beliefs, but others are likely to reject the story as untrue."); Jane E. Baron, Resistance to Stories, <u>67 S. Cal. L. Rev. 255, 256</u>, <u>263 (1994)</u> ("Background assumptions determine, in great measure, whether a particular account will be heard as a ... persuasive or believable story").

³⁶ Just as credibility is very culturally driven, so is relevancy. When we say that one fact proves another and therefore is relevant to show a particular proposition, we are relying on implicit assumptions. See David A. Binder & Paul Bergman, Fact Investigation: From Hypothesis to Proof 178 (1984). Usually these assumptions are embedded with cultural norms, beliefs, and values.

³⁷ Similarly, the lawyer who insists that clients come on time may be surprised to learn that her client considers herself to be on time when the client arrives "late" according to the lawyer's time orientation to minutes and hours. Once a lawyer knows that a client has this relationship to time, the lawyer can schedule appointments to coincide with the client's sense of time by blocking out interview time later than that agreed upon with the client. Other suggestions include talking to the client about the importance of coming at the precise time rather than "on time."

³⁸ Professor Montoya reinforces this point in describing a class on translation given by Professor Zuni Cruz. Professor Zuni Cruz invited a federally-certified Navajo translator to assist in teaching students how to work with translators. "Ms. Yazzie's presentation debunked for all of us the idea that languages are transparent or that representations of reality somehow exist apart from language. One of several examples cited by Ms. Yazzie involved different conceptualizations of time: 'February' translated into Navajo as "the time when the baby eagles are born. Certainly, this is a temporal concept more connected to nature and to place than a word such as 'February' and, as such, is a different construct." Montoya, supra note 3, at 160.

³⁹ Intercultural Interactions, supra note 13, at 302.

solutions, and determine who are important players in a decision. ⁴⁰ For example, in analyzing the scenario described at the beginning of this article, students have very different interpretations as to why the brother might be there, based in part on whether the student sees the brother and sister as a unit or as separate individuals, one of whom has a legal problem. This assessment is very much related to the student's culture and family experience.

Students who explore differences in individual and collective cultures may come to appreciate different communication styles, values and views of the roles of the lawyer and client. In an individualistic culture, people are socialized to have individual goals and are praised for achieving these goals. They are encouraged to make their own [*46] plans and "do their own thing." ⁴² Individualists need to assert themselves and do not find competition threatening. By contrast, in a collective culture, people are socialized to think in terms of the group, to work for the betterment of the group, and to integrate individual and group goals. Collectivists use group membership to predict behavior. Because collectivists are accepted for who they are and accordingly feel less need to talk, silence plays a more important role in their communication style.

Majority culture in the United States has been identified as the most individualistic culture in the world. ⁴³ Our legal culture reflects this commitment to individualism. For example, ethical rules of confidentiality and conflict of interests often require a lawyer to communicate with an individual client in private and may prohibit the lawyer from representing the group or taking group concerns into account. ⁴⁴ In addition, the Anglo-American legal system creates substantive laws that reflect a highly individualistic model of rights and responsibilities. ⁴⁵ Students trained under this system need to be alert to potential conflicts that may arise between a client's culture and the legal strategy designed for an adversarial, individualistic system. Students who understand this are better able to address the problems it creates for those clients who come from or embrace a more collective culture. ⁴⁶ [*47]

⁴¹ See infra notes 196-99 and accompanying text for a classroom discussion of this scenario.

⁴² Intercultural Interactions, supra note 13, at 302.

⁴³ G. Hofstede, Culture's Consequences: International Differences in Work Related Values (1980); G. Hofstede, Cultures and Organizations: Software of the Mind, in Intercultural Interactions, supra note 13, at 302. Other nations that rank high in this dimension are Australia, Canada, Great Britain, the Netherlands, and New Zealand. Nations that score high on collectivism are primarily those in Asia and South America.

⁴⁴ See also O'Leary, supra note 4, at 72. Professor O'Leary points to both the ethical rules and concepts of standing as limiting lawyers' conceptions about who is involved in a dispute. Following Jean's and my presentation at the 2000 AALS Clinical Teacher's conference, Professor Peter Joy alerted us to a contemplated change in California professional responsibility rules on confidentiality, which would allow the privilege to be maintained when family members or others are part of the interview process.

⁴⁵ In her article, Professor Zuni Cruz identifies the differences in an adversarial system and a traditional indigenous system. Zuni Cruz, supra note 4, at 594. Professor Zuni Cruz's article identifies the different role choices that lawyers who practice in more collective societies may make. She encourages an approach that views the client and her legal problems in the context of the client's own community. This framework (as opposed to the client-centered individualistic framework) allows the lawyer to gather different kinds of information, id. at 575, and to counsel clients differently, id. at 577.

⁴⁶ In developing more culturally appropriate services, professional training often addresses the need to assess the impact of an individualistic culture: "The relentless focus on the self provided by most existing services may be alien and disquieting to persons with cultural values that define the self only in concert with others and perceive autonomy and individualism as undesirable or unnecessary." See R.H. Dana, Multicultural Assessment Perspectives For Professional Psychology 16 (1993). See also Derald Wing Sue, Counseling the Culturally Different 78-81 (1981). Sue points out the problems for counselors who incorrectly assume that all clients will respond to a belief that they are responsible for their own actions and can improve their lot in life through their own efforts. In some cultures, "to be actively self-assertive forecasts adjustment difficulties." Id. at 83.

⁴⁰ Zuni Cruz, supra note 4, at 580-84, tells a number of stories illustrating difference in individualistic and community-focused lawyering and how culture influences the choices that lawyers make.

Students who come from more collectivist cultures may themselves experience some of the cultural dissonance that such clients face. ⁴⁷

Here is an example of how a result that appeared successful can nevertheless be unacceptable when viewed within the context of the client's collective culture. In this case, lawyers negotiated a plea to a misdemeanor assault with probation for a battered Chinese woman who had killed her husband and who faced a 25-year sentence if convicted of murder. The client, who had a strong self-defense claim, refused to plead to the misdemeanor charge because she did not want to humiliate herself, her ancestors, her children and their children by acknowledging responsibility for the killing. Her attorneys did not fully comprehend the concept of shame that the client would experience from such a plea until the client was able to explain that the possibility of 25 years in jail was far less offensive than the certain shame that would be experienced by her family (past, present and future) if she pled guilty. These negative reactions to what the lawyers initially viewed as an excellent result allowed the lawyers to examine the meaning of pleas, family, responsibility and consequences within a collective cultural context that was far different than their own. ⁴⁸

In another case, lawyers had to change their strategy for presentation of evidence to make a claim that honored the cultural and religious norms of their client. In this case, lawyers arguing for political asylum for a female client wanted to present evidence of persecution by showing an injury to an area of her body that the client was committed, by religion and culture, to keeping private. Ultimately, the client developed a strategy of showing the injury to the INS lawyer who was also female. ⁴⁹ This strategy, challenging conventional legal [*48] advocacy and violating cultural norms of the adversarial system, allowed the client to present the case in a way that honored her values and norms. ⁵⁰

Each of these cases presented stark cultural contrasts with clear connections to lawyering choices. In hindsight, it is easy to see the cultural contrasts and their effects on the clients' and lawyers' perceptions of what actions were appropriate and what accommodations were acceptable. In the heat of the moment, however, cases are more difficult, and the differences and similarities are more subtle and, at times, invisible. As clinical teachers, our job is to develop ways to make the invisible less so. The next section identifies some critical cross-cultural frameworks and vocabulary for giving our students a way of talking about differences and similarities, so that students will leave our programs better prepared to be lawyers in a multi-cultural world.

II. Developing Cross-Cultural Competence By Articulating Cross-Cultural And Learning Theories

⁴⁹ Peter Margulies, Re-Framing Empathy in Clinical Legal Education, <u>5 Clin. L. Rev. 605 (1999)</u>. Peter also presented this case at the 1999 CUNY Conference, "Enriching Legal Education for the 21[su'st'] Century, Integrating Immigrant's Perspectives Throughout the Curriculum and Connecting with Immigrant Communities."

⁵⁰ The actual fact finder, the judge, never saw the evidence. The adversary learned about the evidence not from the lawyer, but from the client; the adversary, not the advocate, presented the evidence to the court.

⁴⁷ One wonders, for example, what conflicts students who are raised in cultures that value avoiding confrontation and maintaining harmonious relationships face when dealing with an American legal system that fosters competition and zealous advocacy with a winner-take-all ethic.

⁴⁸ This scenario was told to me by Professor Holly Maguigan who for years has represented battered women in criminal cases. In this case, her students worked with a lawyer from the Legal Aid Society. These lawyers were significantly aided by the advocates of the New York Asian Women's Center who perform both language and cultural translations. The Center is a community-based organization that works with a diverse group of Asian women in assisting them to deal with issues of intimate violence. For a more detailed analysis of the difference between individualism and collectivism, see note 45 supra and accompanying text, as well as Handbook of Intercultural Training, supra note 13, at 19.

A starting point for faculty who are designing cross-cultural learning curricula is to ask what students already know and what we want students to learn. ⁵¹ Cross-cultural learning takes place in three different spheres: the cognitive, behavioral and emotional. In planning a class, teachers need to set goals for each of these spheres. We need to question our students' cognitive knowledge and awareness so as to determine what they need to know about cultural style differences, the dynamics of prejudice, and the nature of oppression. ⁵² We need to evaluate the skills that our students have or need to acquire in order to be competent. And lastly, we need to gauge students' motivation to learn and emotional resistance to change and assess what we can do to open students up to learning in this area.

The teaching choices to be made to accomplish change in each of these spheres depends in large part on the assessments we make about [*49] our students, about our capacity to teach these perspectives and skills, and about the connection of these skills to our vision of good lawyering. In addition, we need an explicit theory about what we mean by cross-cultural lawyering competence and an explicit pedagogical theory about how students can learn to be competent.

The Habits, the student materials and the teaching suggestions that are developed in Sections Three and Four are based on Jean's and my theories about cross-cultural competence and learning theory. In the student materials, we lay out the following four guiding principles about the Habits that express some underlying assumptions about good lawyering and learning:

. all lawyering is cross cultural; 53

. a non-judgmental approach towards yourself and client promotes learning ⁵⁴ and good lawyering;

. remaining present with the individual client is an essential part of cross-cultural competence; ⁵⁵ and

⁵³ Lawyering is often cross-cultural and therefore all lawyers need cross-cultural skills. Lawyers who explicitly examine the cross-cultural issues in a case will increase client trust, improve communication and enhance problem solving on behalf of clients." Jean Koh Peters & Susan Bryant, Cross/Cultural Lawyering Materials (2001) (on file with author).

⁵⁴ Refraining from judgments and being open to difference is an essential skill for effective cross-cultural lawyers. We believe that openness is a skill that can be learned and applied towards clients and ourselves. Inevitably, we will continue to interpret behavior through our cultural lens. To honestly unearth our own cultural assumptions, stereotypes and biases and examine them, we need to view them without shame or judgment or self-condemnation, but with an eye towards understanding them and, where necessary, rectifying or eradicating them. To understand our clients, we need to use the same kind of non-judgmental approach. Id.

⁵⁵ Remaining present with the individual client, ever respecting her dignity, voice and story allows lawyers to avoid stereotyping. This principle, a goal of all lawyering, is especially difficult to attain in high-pressure, high-volume practices, where the "efficiency" of categorizing and generalizing, and severe time and resource constraints can lead the lawyer away from an individualized understanding of each client. Especially when we are studying culture and the ways our clients may be socialized by their cultures, we need to remember that although we are all influenced by culture, we are also individuals who may or may not embrace all of the cultural values of our socialization process. Id.

⁵¹ In the training world, this would be called a needs assessment. In conducting such an assessment, a teacher should strive to avoid viewing the students as a monolith, and to keep in mind that differences exist even in the most seemingly homogeneous classrooms.

⁵² For example, a clinic that placed greater emphasis on changing the legal system would devote greater attention to teaching about the nature of oppression whereas one focused on improving representation might focus on cultural style differences. Of course, the two go together, but the focus or emphasis might be different depending on the primary goal. Bernardo M. Ferdman & Sari Einy Brody, Models of Diversity Training, in Handbook of Intercultural Training, supra note 13, at 292.

. knowing yourself as a cultural being is an on-going and necessary process for cross-cultural competence. ⁵⁶

In the following section, I identify other ideas about cross-cultural competence and learning theory that inform our work.

[*50]

A. Cross-Cultural Competence

1. Cross-Cultural Theory

To increase cross-cultural competence, teachers need to plan classes and supervision that teach cross-cultural theory as well as skills. In terms of cognitive goals, teachers should identify concretely the awareness and knowledge goals that build cross-cultural competence. For example, goals might include increasing students' awareness of the significant role culture plays in: ⁵⁷

. giving meaning to behavior and words;

- . developing values and judgments;
- . forming relationships with others; ⁵⁸ and
- . developing biases and stereotypes. 59

In setting knowledge goals, the teacher should identify culture-general and culture-specific information that is important to the students' clinical work and future learning. Culture-general knowledge helps the lawyer understand differences and similarities by means of concepts and experiences that are likely to occur in any culture. ⁶⁰ These concepts will also help students gather specific information on the individual cultures about which they are learning. For example, many of the categories discussed in Section One illustrate cultural concepts that reflect culture-general information related to our work as lawyers. The concepts that seem particularly key include:

⁵⁷ Students need to understand that "crossing cultural boundaries is a two-way process." See Zuni Cruz, supra note 4, at 567-68. The lawyer must consider his own and the client's world views, understanding the client's goals as well as how the lawyer's perspective influences the way the lawyer views the other culture. See also Jacobs, supra note 4, at 405-07, reporting that the cross-culturally competent therapist should be aware of his/her biases and values, and how these might affect clients of color. In terms of knowledge, the counselor should know the role that cultural racism plays in the identity and world views among minority groups. Id., citing Sue et al., Cross-Cultural Counseling Competencies, in 10:2 The Counseling Psychologist 45-51 (1982).

⁵⁸ elip>becoming effective overseas involved a heavy measure of self-understanding and awareness." Intercultural Sourcebook 3-5 (D.S. Hoopes & Venturas eds., 1979).

⁵⁹ See Isabelle Gunning, Diversity Issues in Mediation: Controlling Negative Cultural Myths, <u>1995 J. Disp. Resol. 55.</u> To show how cultural myths can change the way we interpret narratives, Professor Gunning uses a process of race-shifting to illustrate how, if we assign different races to the participants, we view the participant's stories differently. She points out that the cultural myths that will be applied to minority and disadvantaged group members will almost always be negative and rely upon taboos inculcated during childhood and affirmed by media rather than based on personal experience. <u>Id. at 79.</u>

⁶⁰ Id. at 86.

⁵⁶ Knowing ourselves as cultural beings is key to being able to identify when we are using biases or stereotypes, when we are misinterpreting or filling in, and why we are judging people who are different. The lawyer must also accept that his or her culture may create roadblocks to understanding others. Our experiences and our cultures create strong categories of in-group and out-group and cause us to stereotype the "other." By accepting the blinders that shape our understandings of others, we can feel less frustrated by setbacks and not judge ourselves too harshly for having prejudices and biases, as long as we are committed to growth and change. Over time, a lawyer can learn to befriend herself as a cultural being, through self-understanding. Lawyering is an ongoing, lifelong process. Id.

. collective and individualistic cultures; ⁶¹

. time; 62

[*51] . direct and indirect communication;

- . social role and hierarchy;
- . insider-outsider;
- . categorization; and
- . attribution. 63

In identifying culture-specific cognitive goals, the teacher should start by identifying differences and similarities between a student and the client group in the particular culture that the student is learning. The teacher should identify culture-specific knowledge, politics, geography, and history, ⁶⁴ especially information that might shed light on the clients' legal issues. ⁶⁵

In addition to culture-general and culture-specific knowledge, [*52] cognitive goals for students should include understanding two key concepts that will affect their work: (1) the importance of understanding similarities as well

⁶¹ Professor Zuni Cruz details the many difficulties that arise when considerations of community clash with a profession "long cast in terms of individuality." Zuni Cruz, supra note 4, at 568.

⁶² Majority culture in the United States is monochronic; punctuality and adherence to schedule are important. Other cultures (including some within the United States) are polychronic, stressing the completion of tasks and involvement of people rather than strict adherence to schedule. See Kenneth Shipley, Working with Linguistically and Culturally Diverse Clients, in Kenneth B. Shipley, Interviewing and Counseling in Communicative Disorders: Principles and Procedures (2d ed. 1996).

⁶³ Kenneth Cushner & Dan Landis, The Intercultural Sensitizer, in Handbook of Intercultural Training, supra note 13, at 189, provide an extensive list of concepts that reflect culture general information that is relevant to our work as lawyers. They divide these concepts into three categories: people's experiences that engage the emotions, knowledge areas that people learn as a result of being socialized within a culture, and bases of cultural differences influencing how people learn to process information. Within the first category are anxiety about appropriate behavior, disconfirmed expectations, outsider status, tolerance of ambiguity, and confrontation of one's own prejudices. Within the second category fall communication and language use, individualism versus collectivism, rituals versus superstitions, social hierarchies, values, differences in work, and differences in time and spatial orientation. Finally, the last category encompasses the ways in which people categorize information, the differentiations they make between categories, ingroup-outgroup distinctions, learning styles, and attributions.

Several of the cross-cultural training books have 3-4 page descriptions of these cultural concepts that could be used to teach them. Also, teachers can lecture or use students' cases to teach concepts. See Montoya, supra note 4. Obviously language difference is a significant difference that affects our concepts of insider-outsider, categorization and attribution. For example, Sharon Hom describes the genesis of her English-Chinese Lexicon on Women and the Law as a response to the difficulty of translating concepts such as empowerment, sex, gender, feminism, and affirmative action into Chinese. See Sharon K. Hom, Lexicon Dreams and Chinese Rock and Roll: Thoughts on Culture, Language, and Translation as Strategies of Resistance and Reconstruction, <u>53 U. Miami L. Rev. 1003, 1009-11 (1999)</u>. The materials and this article do not address the impact of language difference and how to teach about it. Although this is a very significant difference (and one to which I devote at least one class session per semester), everyone has to stop somewhere. Other clinical faculty have focused on these issues. See Ileana Dominguez-Urban, The Messenger as the Medium of Communication: The Use of Interpreters in Mediation, <u>1997 J. Disp.</u> Resol. 1; Angela McCaffrey, Don't Get Lost in Translation: Teaching Law Students To Work With Language Interpreters, <u>6 Clin.</u> L. Rev. <u>347 (2000)</u>.

⁶⁴ Intercultural Interactions, supra note 13, at 3. Even if the teacher is not teaching culture-specific information, the student should be encouraged to do some of that learning on her own.

⁶⁵ Zuni Cruz, supra note 4, at 593, identifies a number of classes that are focused on culture-specific instruction in native culture and its relationship to lawyering skills and perspectives.

as difference and (2) the recognition that not all similarities and differences have equal significance. The insight about the importance of similarities focused our early work as we recognized the importance of students seeing themselves connected to clients by their similarities. ⁶⁶ If students are encouraged to articulate the similarities in their clients and themselves, in their clients' choices and their own, and in their clients' values and their own, the students may be better able to understand their clients ⁶⁷ and less likely to judge clients harshly. For example, in one class in which students were asked to identify similarities and differences, one student noted that he had been able to list many differences and only one similarity, but that prior to this exercise, he had not seen the similarity. The similarity that he noted was that both he and his client loved their mothers and were very attached to them. This insight was an important connection for the student, who was having difficulty understanding his client, a child in a child protective case. ⁶⁸

We also recognized that students who saw themselves as very similar to their clients often missed differences and made assumptions about client motivations and goals. For example, a student whose parents came from the Dominican Republic, in describing a Dominican client with whom she was working, stated that she saw the client exactly like herself except for differences in their educational backgrounds. The student had spent significant time with the client telling her that she needed to go back to school. This exercise gave the student [*53] insight into why she was giving her client that advice. ⁶⁹

In teaching students some theory about the importance of culture in their interactions and in their attribution of meaning, we wanted students to see that people raised in different cultures can share similarities that are manifested differently. Assumptions of similarity can also create difficulties for students if shades of difference are ignored. For example, all societies have some form of social hierarchy. In one culture, this may be demonstrated by the person of lower rank kissing the feet of a superior; in another culture, this might be accomplished through bowing. In the United States, we frequently acknowledge social status through titles. Although all students have experienced some sort of hierarchy in their own culture, they do not necessarily have a full understanding of ways in which hierarchy is manifested in another culture. By acknowledging both similarity and difference, a student is more likely to explore the client's concept of hierarchy and understand the client's reaction to issues of hierarchy in the attorney-client relationship as well as the client's relationship with others.⁷⁰

Although the Habits and materials ask students to identify a range of similarities and differences, an important component of cross-cultural competence is to recognize that not all similarities and differences have the same

⁶⁷ Professor Judy Scales-Trent describes the importance of recognizing sameness as moving toward a capacity to see the world from multiple perspectives. Judy Scales-Trent, Sameness and Difference In the Law School Classroom: Working at the Crossroads, 4 Yale J. L. & Feminism 415 (1992). Citing Gloria Anzaldua, the focus on sameness as well as difference allows the student to "at once see through serpent and eagle eyes." Id. at 417 n.3. This capacity may encourage students to represent clients who "earlier they had considered unsympathetic (too different)" in a different way with a greater capacity to understand and believe their stories. As a result, students may be more willing and able to investigate and spot issues. Id. at 436.

⁶⁸ See also Aiken, supra note 4, at 39-40.

⁶⁹ This prompted a wonderful and interesting classroom conversation about whether greater distance was necessary in order to be helpful. Was a lawyer even supposed to talk about education? A student who had spent a year working in Guatemala talked about how legal workers in that country doubled as job counselors because of the absence of job counselors. This caused the students to ask if the situation is any different in this country when lawyers are working in under-resourced communities.

⁷⁰ Bhawuk & Triandis, supra note 28, at 23. Cross cultural trainers refer to this kind of similarity and difference as etics and emics. An etic is a principle or generalization that cuts across cultures. For example, hierarchy can be found in all cultures and

⁶⁶ This insight has been the subject of presentations at clinical teachers' conferences as well. For example, I recall Katherine Klein's and Margaret Barry's presentation at an AALS annual meeting demonstrating a method of establishing connection and lessening judgments of clients by getting students to identify situations where they reacted similarly to clients. In our battered women's clinic, during one session on teaching empathy, we asked each student to identify a situation she or he thought was bad but stayed in anyway. We also asked each student to identify a situation where she or he was hurtful to someone she or he cared about. By doing this exercise early in the semester, we were trying to create connections to battered women who often make choices to stay for all the reasons that the students articulated in the class session, and to foster students' understanding of why they stayed, and of why women may be abusive to their children.

degree of importance in all settings. Some similarities and differences might trigger stereotyped thinking that could harm clients. As Michelle Jacobs warned in her article, People from the Footnotes: The Missing Element in Client-Centered Counseling, ⁷¹ the legal profession's failure to address issues of race may result in clients of color being further marginalized by white lawyers. Professor Jacobs identifies ways in which bias and stereotyping may cause [*54] lawyers to misinterpret clients ⁷² or, worse, to provide less service to the poor and clients of color. ⁷³

Similarly, as Jane Aiken recognized in her article, Striving To Teach Justice, Fairness and Morality, ⁷⁴ students need to learn about the concepts of power and privilege and their influence on the attorney-client relationship and the legal system. In an article commenting on Professor Aiken's, Margaret Montoya points out that the analysis of the significance of similarity and difference changes in different contexts. ⁷⁵ Without denying the power of racism, homophobia, sexism and other "isms," Professor Montoya points out that allocations of [*55] "power" and "privilege" can turn on relationships and audience. For example, the importance of similarities and differences will vary depending on, among other factors, the client, lawyer, judge, and kind of case. ⁷⁶

therefore is an etic. By focusing on etics, trainers establish similarities between cultures. An emic is the manifestation of the etic in the individual culture. For example the etic of hierarchy is manifested in India by kissing feet, in Japan by bowing, and in the United States by giving someone a title. Some anthropologists suggest that cultural general training is invalid, that each culture has to be studied for its own uniqueness, and that etics are not valuable. Id. at 23.

⁷¹ Jacobs, supra note 4. "Caution not to stereotype by attributing group preferences to individuals" is one of four key requirements of communication across cultures. Cochran et al., supra note 4, at 204.

⁷² For example, Professor Jacobs identifies ways that a client might be perceived as difficult when instead, the client is reacting to body language from the attorney, or the client is using tactics of resistance to the lawyer's unconscious racism or to an unfair legal system. Jacobs, supra note 4.

73 In analogizing research done in the medical field showing disparate treatment by doctors for African-American patients, Professor Jacobs explores whether lawyers may also unknowingly be engaging in disparate advocacy. Jacobs, supra note 4, at 381 n.144, citing Ditto & Hilton, Expectancy Process in the Health Care Interaction Sequence, 46 J. Social Issues 97, 97-124 (1990). See also Medical School Students Produce Different Diagnoses of White and Black Patients, 29 J. Blacks in Higher Education 38 (2000), citing Saif S. Rathore et al., The Effects of Patient Sex and Race on Medical Students' Ratings of Quality of Life, Am. J. Medicine (2000), concluding that a patient's race and gender influenced medical students' diagnoses. A recent report of racial disparities in sentencing of juveniles lends credence to Jacobs' analysis that lawyers may be engaging in disparate treatment of clients of color. See Fox Butterfield, Racial Disparities Seen as Pervasive in Juvenile Justice, N.Y. Times, April 26, 2000, at A1, summarizing a Justice Department-sponsored study, showing that Black and Hispanic youths are treated with more severity than white youths charged with comparable crimes at each stage of the juvenile justice system. According to the study, black youth make up only 15% of the total population under 18 years of age. However, 26% of black youth are arrested, 31% are referred to juvenile court, 44% are detained prior to trial, 34% are formally processed by juvenile courts, 32% are found guilty, 46% are waived to adult criminal court, 40% are placed in juvenile prisons, and 58% are placed in adult state prisons. See And Justice For Some: A report prepared by National Council on Crime and Delinquency researchers for the Building Block for Youth Project, available at http://www.buildingblocksforyouth.org. Another study of the child neglect system in New York City shows that African-American children are twice as likely to be removed following confirmed reports of abuse and neglect than white children and once in foster care they are significantly more likely to remain in care for longer periods. See Race, Bias & Power in Child Welfare, 3 Child Welfare Watch 1, Spring/Summer 1998. See also Karen Howze, The Cultural Context in Abuse and Neglect Practice for Judges and Attorneys (1996) for a presentation of the ways that cultural context may influences judges and lawyers. Finally, Professor Jacobs points to studies which detail the effect of race on black students with high cultural mistrust who are more likely to drop out of counseling and less likely to share confidential information. This has analogous implications for the attorney-client relationship. See Jacobs, supra note 4, at 345.

⁷⁴ By focusing on teaching students about privilege, Professor Aiken gives students a lens for examining racism and other biases and stereotyped thinking and actions that address these social injustices. Aiken, supra note 4.

⁷⁵ Professor Jacobs also recognizes that power differentials can affect the degree of trust that clients have in lawyers who can use the power differential to manipulate clients. Jacobs, supra note 4.

⁷⁶ Habit Two illustrates a way that students can identify the relevant players for focus on difference and similarity. Habit Two recognizes that understanding the cultural context of all the players is necessary for the lawyer to assess the impact of culture on the case.

Thus, a competent cross-cultural lawyer acknowledges racism, power, privilege and stereotyped thinking as influencing her interactions with clients and case planning, and works to lessen the effect of these pernicious influences. A cross-cultural framework that asks students to look at a variety of similarities and differences allows those students to examine the "isms" and power issues from a different perspective. The cross-cultural perspective helps explain why we use stereotypes and think in ethnocentric ways as well as identifying new ways of thinking and behaving.

2. Cross-Cultural Lawyering Skills

In addition to awareness and knowledge, students need analytical and communication skills to allow them to engage in cross-cultural interactions competently.⁷⁷ Intercultural communication skills include deep listening skills ⁷⁸ and capacities to focus on content rather than style, ⁷⁹ the ability to read verbal and non-verbal behavior, and the ability to adapt conversation management behaviors and style. These are communication skills that lawyers need in every situation and more so in cross-cultural situations. Students may also want to focus on skills that are valued in the client's culture. ⁸⁰

[*56] Cross-cultural analytical skills require capacities to identify assumptions and to make judgments based on facts rather than stereotypes and bias. ⁸¹ Of course, all good lawyering requires the capacity to identify assumptions and recognize those situations in which one is merely filling in the blanks. In the cross-cultural context, this capacity is even more important because the meaning we attribute to client behavior and words is often based on ethnocentric interpretations and, therefore, will often be incorrect. A cross-cultural training theorist has identified the primary goal of effective cross-cultural interaction as developing the capacity to make "isomorphic attributions," i.e., to attribute the same meaning to behavior and words that the person intended to convey. ⁸² A capacity to make isomorphic attributions requires an antenna focused on the differing connotations a word or act may have in the different worlds inhabited by the student and the client. To attribute meaning correctly, students need to be able to imagine multiple possible meanings for behavior and to be flexible and adaptable.

Most importantly, non-judgmental thinking is required to develop connection to and understanding of clients. A cross-cultural anthropologist has referred to this as the capacity to enter the cultural imagination of another, or as "perceiving as normal things that at first seem bizarre or strange." ⁸³ Remaining non-judgmental is a core cross-cultural skill and one that is particularly difficult for lawyers. Our training includes being called upon in

⁷⁷ Norman G. Dinges & Kathleen D. Baldwin, Intercultural Competence: A Research Perspective, in Handbook Of Intercultural Training, supra note 13, at 115. A list of skills that are necessary for effective cross-cultural communication include dealing effectively with conflict, being flexible and adaptable, possessing the skills that support the change process and the courage to speak out, working to remove barriers, and setting challenging but realistic expectations.

⁷⁸ Enhanced listening skills are one of fourteen cross-cultural skills identified by J.H. Katz & F.A. Miller, Skills for Working in Culturally Diverse Organizations, in OD Practitioner 25, 32-33 (1993), cited in Handbook Of Intercultural Training, supra note 13, at 293.

⁷⁹ To improve listening skills, cross-cultural trainers advise a focus on content rather than style of speech. Id. Professor Peters also points out that voice quality, tone, and loudness are culturally influenced verbal cues that we use for interpretation of meaning and evaluation of credibility. See Cochran et al., supra note 4, at 212. Thus, a capacity to separate content from style helps the listener hear and judge differently.

⁸⁰ The students may not be able to acquire all the skills fully, but being able to identify them and work on acquiring them is an important aspect of cross-cultural awareness. For example, awareness that the skill valued by the culture includes listening or facilitating more than dynamic speaking might well cause a lawyer working within this culture to focus on developing these skills.

⁸¹ Characteristics for cross-cultural competency for psychotherapists included avoiding prejudices, labeling and stereotyping and steering clear of limiting notions about minority clients. Jacobs, supra note 4, at 345 n.294.

⁸² Harry Charalambos Triandis, Interpersonal Behavior (1977).

⁸³ Carroll, supra note 17, at 2.

classroom discussion to judge a case based on limited, digested casebook facts. Early in most representations, lawyers begin to assess clients' stories and the likely result in the case. Students are often taught that assessing client credibility is a critical piece of the lawyers' role that begins in the initial interview. ⁸⁴ Although assessing the viability of a case is an important lawyering skill, cultural differences or incorrect assumptions of similarity may lead to ethnocentric distortion of the lawyer's assessments.

The capacity for reflection and debriefing is an essential skill for exploring cross-cultural difficulties and learning cross-cultural knowledge and skills. Effective cross-cultural interaction depends on the lawyer's capacity to self-monitor his or her interactions in order to compensate for bias or stereotyped thinking and to learn from mistakes. Students with a capacity to talk about issues of difference will [*57] be better able to reflect with and learn from others.

Finally, in addition to identifying the cognitive and skill goals, teachers need to take into account the emotional needs of cross-cultural learners. Students need motivation to learn cross-cultural competence, capacity to live with conflict, ⁸⁵ and coping skills to manage the stress that comes from intercultural interactions. ⁸⁶ Positive educational experiences in law school should promote a willingness to continue to explore and learn about similarities and differences and to seek assistance from others to continue to learn. The teacher's structure of the learning environment will influence the student's motivation and capacity to learn these skills and to gain the knowledge and awareness to become a competent cross-cultural lawyer.

B. Pedagogical Theory

Early in our development and teaching of the Habits, we tried to address some of the problems we had seen in prior diversity training sessions that we had attended or given. Jean called these problems "The Three Ghosts of Diversity Past." The first ghost that we wanted to exorcise was the ghost of designing training exclusively for white students. We recognized that intercultural learning was important for all students. Although students of color and white students who have lived a bi-cultural life certainly have a much deeper understanding of the role that culture plays in their lives, even these students often need to learn how to translate the insights gained from living a bi-cultural existence into lawyering skills for a multi-cultural world.

The second ghost that we wanted to address was the tendency to create unfair burdens on students of color to educate white students in diversity training programs. These programs often create stress for students of color and waste their precious educational resources by making them "teachers" without the authority and planning opportunities that faculty normally have. Finally, we wanted to eliminate the ghost of judgment that often resulted in naming and blaming or the fear of such occurrences during a diversity training.

Thus, with the Habits, we endeavor to create learning opportunities and approaches to help all students be better cross-cultural lawyers. Cross-cultural theory explains why we think in ethnocentric **[*58]** ways and thus gives students an understanding of how racism, sexism and other bias and stereotyped thinking develop. ⁸⁷ With this understanding, students can begin to develop habits of mind and skills for interacting that encourage them to address the cross-cultural issues in their lawyering. We use a teaching approach that asks students to be

⁸⁴ David A. Binder, Paul Bergman & Susan C. Price, Lawyers as Counselors 192-95 (1991).

⁸⁵ In identifying what skills are needed for working effectively in cross-cultural environments, trainers note the importance of a capacity to deal effectively with conflict and to become more flexible and adaptable. Anthony Patrick Carnevale & Susan C. Stone, The American Mosaic: An In-Depth Report on the Future of Diversity at Work (1995).

⁸⁶ William B. Gudykunst, Ruth M. Guzley & Mitchell R. Hammer, Designing Intercultural Training, in Handbook of Intercultural Training, supra note 13, at 65; Jacobs, supra note 4, at 345.

⁸⁷ Concepts of intercultural communication provide a framework for understanding how and why racism, sexism and other forms of prejudice and discrimination occur; the cultural origins of these patterns, and the ways these are manifested in daily communication and interaction, as well as their more institutionalized forms." R. Michael Paige & Judith N. Martin, Ethics in Intercultural Training, in Handbook of Intercultural Training, supra note 13, at 51. non-judgmental towards themselves, as they inevitably will make mistakes, yet at the same time stresses the importance of lawyering in a way that addresses stereotypes and biases.

We recognized early on that, if we were to educate students to be less judgmental about their clients, we needed to create an environment that was less judgmental towards the students. ⁸⁸ Cross-cultural training theorists have noted the importance of creating supportive learning environments that challenge the learners to address issues of bias and power. The "support/challenge" components are both critical pieces of the overall goals of cross-cultural training. ⁸⁹ Support is important because it lowers resistance to learning and helps students deal with what can often be a very challenging experience. ⁹⁰ Challenge [*59] is important because, as teachers preparing lawyers for practice, we must be careful to educate our students to do no harm. If we allow unchallenged racist, sexist or ethnocentric comments to go unchallenged, our students may in fact do harm to their clients. In addition, in ethnically and racially mixed educational groups, students who are members of oppressed groups may not comfortably accept a learning environment that does not include challenge. ⁹¹

One of the difficulties faced by teachers planning a class or supervision designed to teach cross-cultural topics is the ability to strike the appropriate support/challenge balance for the class as a whole. Teachers have to recognize that different students have different needs and that, as teachers, we also have different needs. ⁹² In planning, teachers can begin to identify the risks to the students and the potential resistance that might occur. Teachers can also plan for challenge.

In thinking about the risks that we ask students to take, we need to acknowledge that intercultural learning is often stressful precisely because it is change-oriented. We are training students to be non-judgmental and to develop

⁸⁹ The support/challenge approach represents new thinking in the cross-cultural training field. See id. at 40-45. The authors trace the history of intercultural training, describing the 60's as employing cognitive university training methodologies; the 70's as using confrontation and "T" group human relations model; and the 80's and 90's as using an Integrated Alternative Learning Model that employs both experiential and cognitive learning approaches. There is a growing interest in looking at power differentials as an important component in intercultural interactions including interactions between the trainer and learners. The support/challenge model has been described as an ethical foundation for training. The ethics of practice in this field are concerned with both the welfare of learners and the welfare of those with whom the learner will come in contact. Id. at 38.

⁹⁰ Professor Okianer Christian Dark recommends a caring relationship with students as the best way to cope with the issues that result from raising diversity issues in the classroom. In responding to a long list of identifiable risks, Professor Dark notes, "The most important point, ... is that the teacher pay attention to his or her relationships with individual students and the class. Investing time and energy into developing a solid, respectful, and approachable relationship with the students in the classroom will put the teacher in the best position to learn." Okianer Christian Dark, Incorporating Issues of Race, Gender, Class, Sexual Orientation, and Disability Into Law School Teaching, <u>32 Willamette L. Rev. 541 (1996)</u>. By working on positive, respectful, and open relationships, the teacher creates a classroom environment in which people will take risks. Professor Montoya suggests that because "voicing difference can be difficult, [we should search] for new metaphors and images" and permit "ourselves and others to get it wrong within the commitment of trying it again." Montoya, supra note 3, at 162. This thought is core to the Habits, especially Habit Five.

⁹¹ See Dark, supra note 90, at 559 n.64. Students of color or women who listen to racist or sexist comments that are not challenged may feel further marginalized by the conversation. Id. at 559.

⁹² In encouraging professors to integrate diversity issues into the classroom, Professor Dark recognizes that teachers without tenure may face greater risks as some students may negatively evaluate teachers raising these issues. Dark, supra note 90, at 557 n.58.

⁸⁸ Many clinical teachers have recognized the importance of creating symmetry between the teacher-student relationship and the student-client relationship. See, e.g., Peter Margulies, The Mother With Poor Judgment and Other Tales of the Unexpected: A Civic Republican View Of Difference and Clinical Legal Education, <u>88 Nw. U. L. Rev. 695 (1994).</u> Cross-cultural trainers recognize the importance of supportive leaders for the creation of a good learning environment. See Paige & Martin, supra note 87, at 53. Other requirements for good learning include: high degree of structure, affiliation among participants and between participants and teachers, task orientation, challenging lessons for learners, learner involvement, and appropriate pace of learning. Id. at 53.

new levels of tolerance, new modes of thinking ⁹³ and valuing as well as new behavior. ⁹⁴ Students may experience this as a threat to their cultural identity. In addition, some students may experience stress because classmates articulate world views that are painful. ⁹⁵ Other students may experience stress because they [*60] have done something that exposed biases that they are embarrassed to acknowledge.

Reduction of the risks associated with cross-cultural learning requires the creation of a supportive environment in which an atmosphere of trust exists among the students and between the teacher and the students. An atmosphere of trust allows students and teachers to lessen resistance, take more risks, and increase learning. An atmosphere of blame and judgment often leads to learner withdrawal, avoidance and ultimately hostility.

To manage risk and encourage learning, teachers can introduce cross-cultural lawyering issues after the group has established itself as a learning unit. ⁹⁶ Teachers can start with aspects of cross-cultural learning that involve lower risks before moving to more experiential learning situations (such as role plays and simulations) that involve greater potential for exposure. ⁹⁷ Also, teachers can have different conversations in supervision than they have in the classroom. Teachers who try to manage the support/challenge balance suggest that supportive challenge should occur when a student says something that is distasteful, insensitive, or biased. ⁹⁸

Even with a supportive environment and an attention to building incremental learning opportunities, teachers need to take into account [*61] - and plan for - the likelihood of resistance. Resistance occurs when students fail to see

⁹³ Additional stress occurs when learning how to learn is substituted for learning facts. Because a student of new cultures cannot learn it all at once, they need a process for incorporating information as part of building cultural competence.

⁹⁴ This way of thinking is a paradigm shift. Intercultural training is change-oriented, requiring new modes of thinking, valuing, and behaving, such as new qualities of tolerance of ambiguities, non-judgmentalism, ethnorelativism regarding cultural differences, both cognitive and behavioral. Human beings are usually ethnocentric; it is not normal to be ethnorelative. Paige & Martin, supra note 87, at 46.

⁹⁵ See Peggy Cooper Davis, Law as Microaggression, <u>98 Yale L.J. 1559, 1565 (1989)</u>. Professor Davis describes the effects of what social science researchers refer to as 'microaggression,' a phenomenon she describes as "subtle, stunning, often automatic, and non-verbal exchanges which are 'put downs' of blacks" by people who often consider themselves non-racist. The cumulative effect of such encounters in the recipients is lowered self-esteem, feelings of incompetence, and unrelieved psychological stress. Id. Analogizing this theory to classroom interactions, students from the dominant culture may articulate world-views that in their opinion are both justifiable and non-discriminatory, but which may nonetheless be perceived by their colleagues as the latest in an endless series of "microaggressions."

⁹⁶ A group that has strong respectful relationships with each other will have a greater ability to listen to each other and to hesitate before making judgments. See Dark, supra note 90, at 14. Some faculty use cross-cultural themes as part of the introduction of the students to one another and introduce the less threatening aspects of cross-cultural learning early in the semester. See Zuni Cruz, supra note 4. Others consciously teach students methods for dealing with differing views such as repeating what their colleague has said before objecting as a method to ensure listening. See Hing, supra note 4, at 1833. In our diversity circle discussion groups at CUNY School of Law which take place once a year with students, staff, and faculty, we tell participants to say "ouch" when the speaker says something offensive and allow the speaker to fully finish before commenting on offensive remarks.

⁹⁷ Professor Pusch emphasizes that "trainers must decide how the participants can be encouraged to take greater ... risks, moving from the less difficult intellectual and emotional demands ... to exploring their own attitudes and behaviors in greater depth." Paige & Martin, supra note 87, at 48, citing M. Pusch, Cross-cultural Training, in Learning Across Cultures 121 (1994); see Hing, supra note 4, at 1831-33, for descriptions of simulations that can be used to introduce students to verbal and non-verbal interactions in cross-cultural settings.

⁹⁸ Professor Hing suggests: "What you said made me uncomfortable. Although you didn't mean it, it could be interpreted a saying....." Hing, supra note 4, at 1833. When a personal attack arises during class discussion, Professor Dark takes the opportunity to point out the underlying assumptions on which the remark is made. See Dark, supra note 90, at 568.

the relevance of cross-cultural instruction or ascribe greater value to learning other skills. ⁹⁹ Some students are able to discuss cross-cultural issues generally, but have difficulty applying them to their own actions or resist seeing cross-cultural issues in their own work. ¹⁰⁰ Finally, some students may express hostility to having to discuss these issues at all. ¹⁰¹

Although we can try to reduce risk by using safer pedagogies, cross-cultural trainers confirm what we know as clinical teachers - that didactic teaching alone will not develop new skills. Experiential learning is necessary. ¹⁰² Cross-cultural trainers recommend sequencing that starts with cognitive development by means of lectures and discussions and then moves through group problem-solving, critical incidents, ¹⁰³ case studies, and ultimately, experiential learning.

[*62] Like the learning of other lawyering skills, learning cross-cultural lawyering skills occurs through incremental learning and by practice and reflection. ¹⁰⁴ Learning is enhanced with the teaching of cross-cultural theory. ¹⁰⁵ Thus, the students who follow the Habits without understanding the underlying rationales will generally be less skillful. A framework and vocabulary will allow students to analyze and discuss possible problems. Theory

⁹⁹ As discussed infra notes 162-65 and accompanying text, a lawyering focus is critical to students' motivation to learning these skills and perspectives.

¹⁰⁰ We all have experienced raising an issue of race or gender difference and receiving a response from a student that denies even the possibility that race or gender differences may be an explanation for different views. At a New York Law School clinical theory workshop, Maria Arias, my colleague at CUNY who is Latina, told about a supervisory session in which two white women students were talking about the need to tell the client to dress differently for court. Maria, who had not thought about raising this issue with the client, asked the students if they thought their differences were related to "race." The students rejected Maria's analysis and became very uncomfortable. Maria recounted the story to illustrate how hard these conversations are even with students who are progressive and with whom the supervisor has a good relationship.

¹⁰¹ Although it is important to try to reach every student and to show respect for all students, a teacher should guard against devoting too much of class time to persuading a hostile student of the relevance of the skills and knowledge. Instead, I try to answer concerns or get others to answer them, move on and offer to discuss issues after class. If the teacher uses a pervasive approach, the student will have many opportunities to see relevance or to be hostile over the course of the semester. Thankfully, I have never had to deal with students with significant and sustained hostile reactions.

¹⁰² Gudykunst, Guzley & Hammer, supra note 86, at 65.

¹⁰³ Critical incidents are short case studies (a couple of paragraphs at most) involving a cross-cultural interaction that results in a culture clash. The behavior in the case study can be interpreted in a variety of ways. Students are given multiple choice alternatives of what the behavior might signify. Explanations follow these critical incidents. See Kenneth Cushner & Richard Brislin, Intercultural Interaction, A Practical Guide 22 (1996). The brother and sister story told at the beginning of the article and discussed infra notes 201-03, 218-20 and accompanying text, is an example of a critical incident. Cross-cultural trainers use critical incidents as ways to engage students in attribution training to teach the lesson that behavior is interpreted through cultural lenses and to teach specific cultural norms of another culture. The Intercultural Interactions book has numerous critical incidents organized around cultural themes. Intercultural Interactions, supra note 13. The "disorienting event" described by Quigley, supra note 6, at 53, and Aiken, supra note 4, at 132, is a kind of critical incident challenging the student's framework and providing alternative explanations for why things occur. Constructing critical incidents that can be used for discussion purposes is a less threatening way to build understanding and can be followed by simulation exercises or real-life experiences.

¹⁰⁴ Reflection is a core part of adult learning, especially for helping students examine their perspectives and experiences that challenge these perspectives. See Quigley, supra note 6, at 55.

¹⁰⁵ Bhawak & Triandis, supra note 28, at 19. In analyzing the difference between novices and experts in intercultural competence, the authors point out that novices have some experience without theory whereas experts have lots of cross-cultural experience and organize information and experiences by theory. Novices learn procedures and follow procedures. As students have more experiences and learn theory, they move from a procedural stage through stages of development until they are at an autonomous stage and are able to perform without following each procedural step. Sophisticated users of cross-cultural knowledge use broad principles to categorize and solve problems.

enhances observation, causing students to gather different data, and gives students techniques for processing that data in a manner that permits new types of problem-solving.¹⁰⁶

Cross-cultural trainers identify stages of developing cross-cultural competence. ¹⁰⁷ Howell has identified four stages of growth for cross-cultural competence. ¹⁰⁸ The first stage is one of "unconscious incompetence." In this stage, students are not aware of what they do not know. Students are insufficiently aware of the impact of culture, fail to recognize difference when they see it, and often overlook cross-cultural blunders when they happen and therefore do not avoid or try to rectify them. In the second stage, referred to as one of "conscious incompetence," students are aware that culture affects the way they attribute meaning and practice law, but they are unable to use culture theory and skills to develop competent interaction with others. In this stage, students are able to recognize blunders but are not able to avoid them. They are aware that they may not be connecting with clients, but do not have the range of skills that are necessary to build a trusting relationship with the client.

The third stage is one of "conscious competence." Students in [*63] this stage function competently in the range of skills necessary for cross-cultural lawyering, but only after they have adopted a mindful approach to using the skills. The Habits are a conscious approach that creates steps that students can follow to explicitly address issues of similarities and difference. By practicing the Habits, students learn skills that can be consciously applied to their interactions with clients. The final stage of development is one of "unconscious competence," in which students unconsciously incorporate the cross-cultural skills and perspectives as in their interactions with clients.

In planning classes and supervisions, we need to assess our students' - and our own - stage of development. We need to determine where we want and expect to end up at the end of our teaching, and how best to get there. Lectures and discussions may suffice for fostering "conscious incompetence," the second stage of development, but more experiential learning is needed in order to proceed to "conscious competence," the third stage, and finally to "unconscious competence," the fourth and final stage.

How much time should a teacher allocate to the skills and knowledge of cross-cultural competence? ¹⁰⁹ Each teacher will answer this question differently depending on the overall goals of that teacher's specific clinic. Cross-cultural trainers are clear that a one-class session may raise awareness of cultural differences, but that true cultural sensitivity can only take place with practice and reflection over time. ¹¹⁰ The Habits described below can be introduced in two classes, but the skills that are needed to fully implement them must be practiced and honed throughout the students' clinical work and professional life after law school. ¹¹¹ Such a pervasive approach generally is favored by those clinicians who have worked on developing these competencies in their students. ¹¹²

¹⁰⁶ See Intercultural Interactions, supra note 13, at 7; Bhawuk & Triandis, supra note 28, at 19. See also Quigley, supra note 6, at 63, alerting the reader to the danger of generalizing from one experience and suggesting class discussions as a way to avoid that. Teaching theory and giving students concrete information about specific cultures is another way to counteract the inaccurate generalization process.

¹⁰⁷ See Paul Pederson, Developing Interculturally Skilled Counselors (1986), describing a developmental model designed by the staff of a National Institute of Mental Health Training Project at the University of Hawaii which uses a three-stage sequence of moving people from awareness to knowledge to skill.

¹⁰⁸ Bryan Adamson introduced Howell's concepts to the clinical community as part of his presentation at the 2000 AALS Clinical Teachers Conference.

¹⁰⁹ See O'Leary, supra note 4. O'Leary advocates making difference analysis a part of the problem solving process that is taught and used throughout the semester rather than relegating it to one or two classes on difference analysis. See also Jacobs, supra note 4, at 406 (recommending that issues of race, gender, class and ethnicity be taught pervasively, including in classes on theory, application and reflection).

¹¹⁰ See Gudykunst et al., supra note 86, at 62.

¹¹¹ Jean and I most often devoted one class to teaching the Habits. We never got through all five. With the exception of Habit Three, we have found that unless we raise the Habits in supervision, students are unlikely to engage in this kind of thinking as a routine matter even after they have had had the class. Our experience confirms that of cross-cultural trainers and what we

[*64]

III. The Five Habits of Cross-Cultural Lawyering

In this section, I summarize the Habits that are described more fully in materials and worksheets written for students. I have placed these materials on a web site that allows the materials to be downloaded and adapted to fit different teachers' teaching goals. The materials are written with a litigation focus because that is the setting in which Jean and I work and our examples come from our clinical and lawyering work.

A. Habit One: Degrees Of Separation And Connection

Mary, an Irish-Catholic thirty-year old student in a domestic violence clinic, submits two client-intake memos. In one she describes an intake interview that she had with Razia, ¹¹³ a recent immigrant from Pakistan. Razia has been subjected to spousal abuse and is preparing to leave the marital home. The memo provides a detailed description of Razia's family and reports that she has intentionally isolated herself from her large extended family which does not support her decision to leave her husband. Razia cannot rely on family members to attend court proceedings with her or to provide any assistance. In another memo which describes an interview with another client, Maureen, who is a recent immigrant from Ireland, Mary reports little information about Maureen's family, other than to report that she has moved in with her sister to escape spousal abuse.

Why does the student have extensive information about one client's family and little about another's? Is it the influence of the questions the lawyer asked or the story the client volunteered?

Habit One gives students a framework within which to analyze these questions regarding how similarities and differences between the lawyer and client may influence lawyer-client interactions, especially information gathering. Habit One first asks students to list and diagram similarities and differences between themselves and their clients and then to explore the significance of these similarities and differences. By asking students to identify differences, we focus them consciously on the possibility that cultural misunderstanding, bias and stereotyping may occur. ¹¹⁴ By focusing on similarities, we make conscious [*65] the connections that students have with clients. ¹¹⁵

We ask students to brainstorm long lists that will enable students to see their clients as individuals who have personal, cultural and social experiences that shape their behavior and communications. ¹¹⁶ We encourage students to make lists honestly and non-judgmentally, thinking about what similarities and differences might affect

know as clinical teachers: learning requires practice, supervision and reflection. See also Jacobs, supra note 4, at 409 (referring to the American Psychological Association's recommendations for cross-cultural competence: "The culturally skilled counselor was one who monitored his functioning through consultation, supervision, and continued education.").

¹¹² Zuni Cruz, supra note 4; O'Leary, supra note 4; Montoya, supra note 4; Quigley, supra note 6; Hing, supra note 4, at 1812; Aiken, supra note 4.

¹¹³ The stories of students and clients introducing the Habits are fictional and based on composites of students and clients.

¹¹⁴ With each client, students may identify different categories that will influence the case and the lawyer-client relationship. These lists will change as the relationship with the client and the client's case changes. Examples of some generic categories include: Ethnicity, Economic Status, Marital Status, Race, Social Status, Role in Family, Gender, Language, Immigration Nationality, Sexual Orientation, Religion, Age, Physical Characteristics, Education, Time Orientation, Individualistic/Collective, Direct or Indirect Communication Style.

¹¹⁵ Articulation of similarities also allows the student and teacher to examine whether assumptions of similarity are accurate and to explore issues of transference.

¹¹⁶ In creating long lists, we do not mean to suggest that all similarities and differences have the same order of importance for students and your clients. For example, in interactions involving people of color and whites, race will likely play a significant role in the interaction. Professor Michelle Jacobs has detailed the importance of recognizing the contextual experience of race

their ability to hear and understand their clients' stories as well as what similarities and differences might affect the clients' willingness to form relationships with them as lawyers.

Another way we teach students to illustrate the degrees of connection and separation between themselves and their clients is through the use of a simple Venn diagram. The graphical representation of Habit One asks students to imagine impressionistically the worlds of lawyer and client as two circles, overlapping broadly if the worlds of the client and lawyer largely coincide and overlapping narrowly if they largely diverge. The diagram often stimulates insights distinct from those gained by making lists. For example, a list of similarities may be small and yet a student may feel very similar to the client because of one shared similarity or, alternatively, the student may have many similarities and yet, for other reasons, find herself feeling very distant from the client. ¹¹⁷

Habit One asks students to look at two effects that may flow from the identification of similarities and differences: first, at the effect on professional distance and second, at the effect on information gathering. ¹¹⁸ Although no perfect degree of separation or connection exists between lawyer and client, teachers often assess relationships between students and clients as "too close" or "too far." ¹¹⁹ Habit One encourages [*66] students who have long lists of similarities or whose circles overlap broadly to ask themselves what, if any, differences they may be overlooking. By pondering this question, students can recognize that even though similarities promote understanding, misunderstanding may flow from an assumption of precise congruence.

Habit One also encourages students who have long lists of differences to question if there are any similarities that they may be overlooking. Negative judgments are more likely to occur when a client or lawyer sees the other as an "outsider." If students identify similarities, they will be less likely to judge their clients negatively. ¹²⁰ Even if negative judgments persist, students can identify their source and ask themselves how to bridge the huge gap between the clients' experiences and their own. Finally, in analyzing the effect of similarities and differences, Habit One allows students to examine ways in which these factors affect clients' senses of closeness and distance to their lawyers.

Habit One also asks students to explore the effects of similarities and differences or assumptions about similarities on questioning and case theory. One example of information gathering occurs when lawyers probe for clarification in interviews. Lawyers usually ask questions based on differences that they perceive between themselves and their clients. ¹²¹ Thus, lawyers tend to ask questions when clients make choices that the lawyers would not have made or when the lawyers perceive an inconsistency between what the clients are saying and doing. Lawyers tend not to ask questions about choices that clients have made when the lawyers would have made the same choices;

and how the failure to do so can impede a lawyer's capacity to understand the client's story or assist other legal decision makers in understanding the client's story. Jacobs, supra note 4, at 372.

¹¹⁷ This allows us to reinforce the point that all similarities and differences are not equal nor are they fixed. See infra notes 158-65 and accompanying text.

¹¹⁸ Habits Four and Five examine other effects that may flow from similarities and differences. Jean and I think of the Habits as a "work in progress" because as we collectively focus on identification and analysis of the effect of differences and similarities, through the use of the Habits, we learn and revise.

¹¹⁹ Distance is itself a cultural concept and its value is culturally determined. The appropriate professional distance is often contested. As teachers, we need to recognize and make conscious our own cultural assumptions about the value and limitations of distance. Although we might draw different lines for appropriate distance, we might all agree that when we speak for the client rather than trying to understand the client or when we judge the client without understanding the client, we lack the requisite degree of professional distance or attachment.

¹²⁰ See supra notes 25-27 and accompanying text on the link between negative judgments and lack of connections.

¹²¹ Cross-cultural trainers explain that the primary way that people with a direct style of communication use to lessen uncertainty in cross-cultural situations is to question where they see differences. Although not attributing it to cross-cultural theory, Professors Isabelle Gunning, Steve Hardwell and I demonstrated this observation in a mock class on interviewing and difference at the AALS Clinical Teachers Conference in 1992. See also O'Leary, supra note 4. in such a situation, the lawyer usually assumes that the clients' thought processes and reasoning are the same as his or her own.

By introducing students to the effects of culture on our view of what we perceive as normal and how it drives questioning, we give students additional tools for evaluating the thoroughness and accuracy of their interviews. For example, perhaps Mary, the student in the introductory example, has different information about each client because she explored the family issues with Razia but left them unexplored [*67] with Maureen. If this occurred, a Habit One analysis could show Mary that these different approaches may be tied to issues of similarity and difference. Mary may have probed the first client about her failure to seek family support because Mary believes that her own family would support her decision to leave an abusive relationship. These differences may arise out of cultural differences in family relationships, assessment of appropriate uses of the law or outsiders to resolve problems, responses to violence, or a number of other explanations. ¹²² Mary leaves unexplored the report that Maureen has moved in with her sister to escape the spousal abuse because Mary imagines taking similar steps if she were in such a situation and implicitly makes a host of assumptions about cultural values that relate to Maureen's family, her family values and her sister and their relationship. The fact that Maureen shares a common ethnic background with Mary may contribute to these assumptions of similarity.

Assumptions of similarity that mask difference can lead the lawyer to solutions and legal theories that may not further the client's goals. Mary may, for example, assume incorrectly that Maureen's sister is supportive of the client's decisions and shares the client's view that moving out is appropriate. This erroneous assumption about family support may cause Mary to neglect to explore other necessary housing arrangements or supportive environments.¹²³

To identify the unexplored cultural assumptions, Habit One encourages students to reflect on the attorney-client interview and identify areas in which they may have missed relevant explanations of behavior. It asks the students to critically evaluate the choices they have made about subjects to explore. Habit One introduces students to a systematic way to name similarities and differences, to examine some of the known effects on the attorney-client interaction from these similarities and differences, and to develop their own insights into the effects of culture on their lawyering. Finally, Habit One cautions students to avoid seeing their clients or themselves or their relationship as fixed by the charts or lists. Culture is dynamic and the importance of different identities change as the situation changes.¹²⁴

[*68]

B. Habit Two: The Three Rings

Habit Two asks students to identify and analyze the possible effects of similarities and differences on the interaction between the client, the legal decision-maker and the lawyer - the three rings. Habit Two asks the student first to identify the differences and similarities between the client and the legal system and between the lawyer and the legal system. After identifying and analyzing this information, Habit Two links this analysis to the Habit One analysis to explore all the ways in which culture may influence a case.

¹²² The parallel universe thinking that is evident here is described more fully in Habit Three. See infra notes 129-31 and accompanying text.

¹²³ Family relationships are incredibly rich areas for cultural misunderstanding and therefore, assumptions of similarity are perhaps even more problematic when issues of family are involved. See Lundy Langston, Political and Social Construction of Families Through Pedagogy in Family Law Classrooms, <u>73 Denv. U. L. Rev. 179 (1995)</u>.

¹²⁴ See Paul P. Petersen & Allen Ivey, Culture-Centered Counseling and Interviewing Skills (1993), describing the self as dialogical, emphasizing cultural relationships, and explaining that one must be aware of the changing patterns in one's culture in relation to others. "Different Aspects become salient at different times<elip> . Culture-centered training is an attempt to develop the capability to manage cultural complexity and respond appropriately to dynamic changes in cultural salience." Id. at 13.

In pinpointing and recording similarities and differences in the legal system-client dyad, students are asked to identify the cultural differences that may lead to different values or biases, causing legal decision-makers ¹²⁵ to negatively judge the client and the similarities that may establish connections and understanding. What does a successful client look like to this decision-maker? How similar or different is the client from this prototype of a successful client? What are the cultural values and norms implicit in the law that will be applied to the client? Does the client share these values and norms or do differences exist?

In focusing on the lawyer-legal system dyad, the student is asked to compare himself or herself to the legal system. To what extent does the student share the values and norms of the law and legal decision-makers? What are the similarities and differences between the student and these decision-makers? To what extent has the student become acculturated to the law and legal culture? How much overlap is there between the student's view of the "successful" client and the views of the law and legal decision-makers?

If helpful, students can use diagrams to chart the differences and similarities as they did in Habit One to visualize different dyads - lawyer/client, client/legal system, lawyer/legal system.¹²⁶

After preparing the diagrams and/or making lists of the three different [*69] dyads, the students are encouraged to search the information for insights about the impact of culture on the attorney-client relationship, the case, and potential successful strategies. In this process students may identify influences that may be invisible, and yet nonetheless affect the case.

To enhance this analysis, students are encouraged to ask the following questions:

Assessing the legal claim: How large is the area of overlap between the client and the law? Do I feel that my client has a relatively weak or a relatively strong claim? In what ways does the legal culture embrace the values and assumptions of my client's culture, her understanding of the problem and the possible solutions? How can I bridge any existing gaps? What additional facts can I use to strengthen the case?

Assessing Credibility: How credible is my client's story? Does it make "sense"? To what extent is knowledge of the client, her values and culture necessary for the story to make sense? How credible is my client? Are there cultural factors in the way the client tells the story that will affect a decision maker's evaluation of my client's credibility? What can I do if those exist?

Legal strategies: Can I shift the law's perspective to encompass more of the client's claim? ¹²⁷ Do my current strategies in the client's case require the law or the client to adjust perspectives? What additional facts or characteristics are needed to strengthen the case? Even if we prevail, will I do harm to my client's larger interests? ¹²⁸

Bones to pick with the law: How large is the area of overlap between the law and myself? Are there points on which I strongly agree or disagree with the law in this area? Do I have an agenda that the client does not have?

¹²⁶ Because Habit Two requires the exploration of three frames of reference, Jean came up with the rings as a way to assess the perspectives and analyze where there was overlap of all three perspectives and where there were differences. Not everyone comfortably uses the diagrams or thinks in the visual ways that diagramming encourages. Habit Two can be done with lists, filled in Venn-diagrams or other imaginative ways that help the lawyer concretely examine the cultural differences and similarities that are involved in a case. The materials for students have samples to assist students in using the diagrams.

¹²⁷ See, e.g., Taryn Goldstein, Should the American Criminal Justice System Recognize a Cultural Defense?, <u>99 Dick. L.</u> <u>Rev. 141 (1994)</u> (analyzing cases that integrated "cultural defenses" into traditional defenses of insanity, mistake of fact, diminished capacity, and self-defense).

¹²⁸ Montoya, supra note 3, at 161.

¹²⁵ We use the term "legal decision maker" rather than judge because there are many players in a legal system who make decisions that will have a significant influence on what happens in a case. For example, a prosecutor, a pre-sentence probation officer, and a judge may all make decisions that influence how the client charged with a crime will be judged and sentenced. Therefore, at various points in the representation, they should be included in the diagram of similarities and differences.

Questions based on assumptions about similarity and difference: How large is the area of overlap between all three circles? Are there areas of inquiry that are important to the legal system but missed by the lawyer? Does my client have a plausible claim that is difficult for me to see because of these differences or similarities? Am I probing for clarity by using three frames of reference - the client's, the legal system's and mine - or am I focused mostly on my own frame?

Hot button issues: Of all the characteristics and perspectives listed on the rings, which loom largest for me? Are they the same ones that [*70] loom largest for the client? For the law?

We have found Habit Two particularly helpful when the students or we are troubled by a case or client. Habit Two analysis helps identify why a focus on a particular aspect of a case can assume preeminence even though that aspect may not be critical to the success of the case. The student can gain insight into why a judge is bothered by a particular issue, or why a client is resisting the lawyer's advice. Students might also begin to understand why clients are prone to view the lawyer as part of a hostile legal system when there is a high degree of overlap between the lawyer and the legal system but only a small degree of overlap between the client and legal system.

The challenge for our students and us is to use these insights to help clients. Lawyers can ask whether the law and legal culture can be changed to legitimate the client, her perspective and her claim. Lawyers can examine whether the current strategies in the client's case require the law or the client to adjust perspectives and to explore whether other choices exist. Can the lawyer push the law or must she persuade the client to adapt? Hopefully, by gaining some of these insights, the lawyer may be better able to translate the client to the legal system and the legal system to the client.

C. Habit Three: Parallel Universes

Mary, a 40-year-old Polish-American, is seeking custody of her 8-year-old child, Alison. Mary has been the primary caretaker for Alison and has not worked outside the home since her oldest daughter was born 15 years ago. Mary's husband George, works as a factory worker in a local plant. Mary's student lawyer, Annette, is a 40-year-old immigrant from the Dominican Republic who came to the United States at age 10. Annette advises Mary that she should seek counseling for her 8-year-old daughter as recommended by the court. Annette has her 12-year-old son in counseling and has found it very useful. Annette tells Mary that the court believes that therapy is necessary to help Alison adjust to the separation of her parents and to improve Alison's relationship with her father. Mary agrees to seek counseling. One month later, Annette learns that Mary has not set up an appointment for Alison with a therapist. Angry, Annette believes that Mary either does not care enough about her case or that she does not credit Annette's assessment that Mary's case will improve if she arranges counseling for Alison.

Habit Three teaches students a method for exploring alternative explanations for clients' behaviors. ¹²⁹ The habit of "parallel universes" [*71] thinking invites students to look for multiple interpretations, especially at times when the student is judging the client negatively. Often when our interpretation of client behavior differs from a student's view of the matter, we ask the student to examine whether there are alternative explanations for the behavior. Habit Three teaches students to ask themselves this question and gives them a cultural framework for analyzing possible explanations for their judgments. ¹³⁰

Students are taught that in a matter of minutes, the lawyer can explore multiple parallel universes to explain any given behavior. In the above example, one can imagine many different explanations for the client's behavior: the

¹²⁹ Parallel universes is a concept from science fiction, which suggests that every reality has multiple alternatives. See, e.g., James P. Hogan, The Proteus Operation (1985); Michael P. Kube-McDowell, Alternities (1988).

¹³⁰ Frequently, if we ask students to brainstorm theories from the perspective of different roles, we see students developing alternative explanatory theories. See, e.g., Nancy Cook, Legal Fictions: Clinical Experiences, Lace Collars and Boundless Stories, 1 Clin. L. Rev. 41 (1994). The cross-cultural framework is one of many frameworks that push students to look for alternative explanatory theories. The value of the perspective of culture is that it alerts students to possible explanations that they may not have recognized and alerts them that they may be using ethnocentric analysis. For example, if the student considers the possibility that the client thinks of time differently and then explores theories that might explain what appear to be

client has never gone to a therapist and is frightened; in the client's experience, only people who are crazy see therapists; the client has no insurance and is unable to pay for therapy; the client cannot accept that the court will ever grant custody to the husband, given that he was not the primary caretaker; or the client did not think that she needed to get her child into therapy immediately, etc. Race and class differences between the lawyer and client may account for Mary's failure to follow her attorney's advice, or her inaction may simply be the product of a tendency to procrastinate.

In another example, the student-lawyer of a client who fails to keep appointments can explore parallel universe explanations for the student's initial judgment that: "My client does not care about the case." Encouraged to think of alternatives, the student may attribute the behavior to a lack of carfare, failure to receive the letter scheduling the appointment, or losing her way to the office. Maybe the client had not done what she promised the lawyer to do before the next appointment or simply forgot about the appointment because of a busy life.

The point of the parallel universe habit is to become accustomed to challenging oneself to identify the many alternatives to the interpretations to which we may be tempted to leap, on insufficient information. By engaging in parallel universe thinking, lawyers are less likely to assume - usually on the basis of limited information - that [*72] they understand the reasons for clients' behavior. Parallel universe thinking also allows the lawyer to follow the advice of cross-cultural trainers that one way to reduce the stress in cross-cultural interactions is to ask "I wonder if there is another piece of information that, if I had it, would help me interpret what is going on?" ¹³¹

D. Habit Four: Pitfalls, Red Flags And Remedies

Jeff, an experienced lawyer who is Jewish and fourth generation Austrian-American interviews his client Charles, an 8-year-old African-American, the subject of a child-neglect proceeding. Charles, who lives in a low-income housing project, has been described by social workers as "a bright, verbal boy." In their first meeting, Jeff gives Charles his standard explanations of "protective hearings," "pleas," and "neglect." In his interactions with Jeff, Charles is subdued and reticent to talk other than saying, "I did not do anything wrong." Thinking of the many children who blame themselves for neglect proceedings against their parents, Jeff explains that neglect proceedings are brought by the state against his parents and not against him.

After a court proceeding that occurs later in the representation, Charles asks Jeff why there were no police in the courtroom. In response to Jeff's question about why Charles thought there would be police, Charles replies, "You only get a lawyer if you've done something wrong." He explains that everyone whom he knows who had a lawyer was put in jail.

How do we encourage students to recognize problem conversations in the moment and plan for corrective steps that avoid some of the problems Jeff and Charles had in this encounter?

The first three habits focus on ways to think like a lawyer, incorporating cross-cultural knowledge into analyzing how we think about cases, our clients and the usefulness of the legal system. Habit Four focuses on cross-cultural communication, identifying some tasks in normal attorney-client interaction that may be particularly problematic in cross-cultural encounters as well as alerting students to signs of communication problems. Habit Four encourages conscious attention to the process of communication - a skill and perspective that clinical teachers have used to improve interviewing skills in all attorney-client interactions. By identifying potential cross-cultural pitfalls and attending to these issues, students will likely perceive aspects of the interaction that they may have missed even if they were paying attention to the process of communication.

[*73] Although Habit Four is a habit that can be done in the moment, beginning lawyers often have difficulty paying attention to both the process and content of an interview at the same time. Habit Four helps them plan for the

temporal inconsistencies, the student may perceive new avenues for investigation. See Binder & Bergman, supra note 36, at 163-65.

¹³¹ Richard Brislin & Tomoko Yoshida, Intercultural Communication Training: An Introduction (1994).

interview. The student will be better able to focus on both content and process by identifying, in advance, indications of good communication as well as "red flags" that show that accurate, genuine communication is probably not occurring.

Habit Four encourages culturally sensitive exchanges with clients, by identifying four areas on which students should focus carefully: (1) scripts, especially those describing the legal process, (2) introductory rituals, (3) client's understanding and (4) culturally specific information about the client's problem.

Habit Four encourages students to use scripts carefully. The more we engage in a particular activity, the more likely we are to have a "script." For lawyers, this often develops into scripts for the opening of interviews, explaining confidentiality, building rapport, explaining the legal system and other topics common to the lawyer's practice. In preparing for interviews, students are encouraged to think about how they will explain some of these concepts. Early in their work as lawyers, students begin to develop the scripts for explanation. Habit Four encourages the student, especially in cross-cultural encounters, to develop a variety of communication strategies to replace scripts. The example of Jeff and Charles demonstrates the problems of relying on scripts and stereotypes. In delivering scripted responses to Charles, Jeff assumes he knows what is bothering Charles because other children have had this reaction. ¹³²

What did Charles mean by "I did not do anything wrong?" Habit Four encourages the student to gather culture-sensitive information through use of a narrative mode. Habit Four asks students to engage in "attentive listening" to the client's story and voice. ¹³³ Habit Four orients the conversation to the client's world, the client's understandings, the client's priorities, and the client's narrative. ¹³⁴ Students are encouraged to ask questions that explore how others who are close to the client might view the problem and how they or she might resolve [*74] it. ¹³⁵ For example, had Jeff explored even briefly the client's response to the situation, Jeff may have learned the source of the client's apprehensions. ¹³⁶ These questions help the lawyer understand the context within which the client sees the problem. ¹³⁷

Finally, if the client has come from another country, Habit Four encourages students to gather information about expectations the client may have about the lawyer and the legal system by learning about how the legal system works in the client's country of origin. These questions also give the student some information about the client's

¹³³ For discussion of the dangers of question mode identified in Habit One, see infra notes 113-24 and accompanying text.

¹³⁴ Although narrative mode may be very useful for the lawyer, this mode may be difficult for people in some cultures. In certain cultures, for example, conversation space is expected to be shared equally and monologues are considered rude or boring. See Carroll, supra note 17, at 23.

¹³⁵ These questions may, for example, help the lawyer appreciate that what she views as an individual concern is, from the client's perspective, a matter involving various other people in the client's life.

¹³⁶ Questions that focus on how the client sees the situation include: What are the client's ideas about the problem? Who else has the client talked to and what advice did they give? What would a good solution look like? What are the most important results? Will anyone other than the client be affected? Consulted? Are there other problems caused by the current problem? Does the client know anybody else who had this problem? How did they solve it? Does the client consider that effective? The potential dangers of question mode are identified in Habit One. However, questions designed to develop a better understanding of how the client sees the situation will assist the lawyer in understanding clients and their objectives.

¹³⁷ I was reminded of the importance of exploring this information in a casual conversation that occurred with a client after the formal legal counseling session was over and we were waiting together while the student went to xerox papers. The client, whose year-long divorce case was coming to a close, nervously asked me whether she would have to live with her husband for 3 months before the divorce could become final - which, she explained, was the way it worked in her home country.

¹³² Knowledge about other children's reactions may be helpful to Jeff in identifying facts and feelings to explore with clients but using that knowledge to make assumptions about how the client is feeling is very risky. Cross-cultural trainers warn that we should not apply generalizations to the individual without confirming their applicability. See Sue, supra note 46, at 43-44.

expectations of the legal system and the lawyer. For example, in many legal cultures, the lawyer is the "fixer" or the person-in-charge. ¹³⁸ In contrast, most law students in the United States are taught a model of client-centered lawyering that calls for a partnership and our ethical rules define a relationship in which the client makes decisions about resolving the case.

Habit Four also encourages a student in a cross-cultural encounter to pay special attention to the beginnings of communications with the client. ¹³⁹ Each culture has introduction rituals or scripts as well as [*75] trust-building exchanges that build rapport and promote conversation. Students are encouraged to consult translators and to pay careful attention to cues from the client in the beginning stages of the interview. For example, students are encouraged to think about what kind of exchange of information early in an interview is likely to build confidence and connection. Students are also asked to explore the different cultural perceptions that lawyers and clients may have about "getting down to business." ¹⁴⁰

Habit Four teaches the student that cross-cultural encounters can generate anxiety for the lawyer and/or the client. One cause for such anxiety is the concern that the other will not understand. Thus, concrete evidence of understanding will lessen that anxiety.¹⁴¹ Habit Four suggests that students can lessen client anxiety by using the active listening technique of providing feedback to the client.¹⁴² Habit Four alerts the student to look for culturally sensitive feedback from the client that the client understands the lawyer or is willing to ask questions about matters that are unclear to the client. Many students are amazed or hurt that a client is unwilling to seek information or clarification. Students may attribute this to a failure of the relationship rather than to a client's communication style.

¹³⁹ Research on initial interviews suggests "the importance of listening well, and with attention for subtle embedded messages, at the very start of an interview, with the expectation that the client will reveal key information." Gay Gellhorn, Law and Language: An Empirically-Based Model for the Opening Moments of Client Interview, <u>4 Clin. L. Rev. 321, 335 (1998).</u>

¹⁴¹ The connection between uncertainty and anxiety is especially close in cultures that engage in uncertainty avoidance as a part of culture. See Brislin & Yoshida, supra note 131, at 22.

¹³⁸ In one cross-cultural lawyering class's discussion of clients' expectations for lawyers, a student from the lvory Coast told the class that in his country the lawyer was the fixer who would solve the problem as the lawyer saw fit whereas here he was being taught to think of the client as a partner and decision-maker. We had a very interesting discussion about the appropriate choice of model for representing clients from the lvory Coast who have these kinds of expectations. The student who gave the illustration argued that people in this country should be expected to work with lawyers in an American style while others talked about trying to meet the clients' expectations and eschewing a fixed style. We considered how one would find out the client's expectations about the lawyer's legal system without having the client think that we did not know what to do. Another complication was dealing with the client's belief that the lawyer could "fix" the situation in contrast to a situation where a judge would decide.

¹⁴⁰ In a conversation I had with Margaret Montoya about this point, she told a story of a videotape she uses with a client named Montoya in which the client and Margaret spend the first part of the interview exploring where her husband's family, the Montoyas, are from and whether there is any family connection between the client and lawyer. In analyzing the tape, her students had very different views about the beginning of the interview. Some students saw this as a waste of time, other students saw it as trust building and the appropriate conversation to have at that point of the interview. Professor Montoya attributed the students' responses to cultural differences in introductory rituals and the importance of building personal connection as well as professional connection. Cross-cultural trainers note the differences in cultures on rituals for beginning conversations. For some cultures, a person who gets right down to business and eliminates the informal and friendly interaction is considered rude or abrupt. See M.E. Zuniga, Families with Latino Roots, in Developing Cross-Cultural Competence: A Guide to Working With Young Children and their Families 151-79 (E.W. Lynch & M.J. Hanson eds., 1998).

¹⁴² Feedback should include both the facts and reasoning. It is less clear how the active listening ideal of identifying the emotional content of the client's communication should be applied to clients from more indirect cultures. One might hypothesize that a client who would be reluctant to directly name the way she is feeling may feel uncomfortable with the lawyer giving feedback on the emotional content of the message.

The lessons of Habit Four ask the student to develop a broader repertoire for determining [*76] client understanding. ¹⁴³

In addition to identifying some potential hot spots in cross-cultural interactions, Habit Four encourages the student to plan for "red flags" and culturally sensitive correctives that will assist the student in paying attention to the process as well as the content of the interview. ¹⁴⁴ These red flags include indications that clients are disengaged, angry, ¹⁴⁵ actively uncomfortable or using the lawyer's terminology. For example, in analyzing Jeff's conduct, students can recognize Charles' reaction during the initial interview of remaining quiet as a possible red flag. Other warning signals include: the student dominates the conversation, has not taken any notes for many minutes, is judging the client negatively or is distracted and bored.

In exploring corrective measures, the main mistake that we often make is to use the same approach to correct the problem that may have caused the problem in the first instance. Habit Four alerts students to the importance of trying a different approach. For example, if the client is not responding to a direct approach, the student might try an indirect approach. If the call for a narrative is not working, the client can be asked some specific questions or asked for a narrative on a different topic. Often the cause of the red flag will be difficult for the student to decipher. By developing a list of corrective measures, ¹⁴⁶ the student will acquire some remedial strategies.

E. Habit Five: The Camel's Back

A woman client, with a horrible story of torture, whom the lawyer had very limited time to prepare for trial (because the client lived out of [*77] town), was talking in a rambling fashion. The lawyer, who was just back from vacation, was thinking angry thoughts about the client. Under the extreme stress caused by time pressure and by listening to the client tell about some horrible rapes that she had suffered, the lawyer was falling back on very disturbing old conditioning: against people who are of a different race, people who are overweight, people who "talk too much." What is the lawyer to do with these reactions?

Many faculty who have looked at the Habits wonder why we do not start with Habit Five which involves exploring oneself as a cultural being. We leave it until last because this Habit can be the most difficult in that it asks the student to face the sometimes ugly side of cultural blinders - bias and stereotype. ¹⁴⁷ Moreover, the other Habits give insight and understanding that may ultimately help students recognize bias. Habit Five proposes two ways to work with bias and stereotype. First, to prevent these factors from affecting the relationship or decisions, Habit Five

¹⁴⁴ Cross-cultural trainers have talked about these red flags as "uncomfortable moments in interviews." Researchers found fewer uncomfortable moments when counselors and clients shared common cultural membership. See Pedersen & Ivey, supra note 124, at 106.

¹⁴⁵ Whether the client is in fact angry or the lawyer is misinterpreting the anger, the lawyer should respond by employing correctives such as asking herself whether there is something the lawyer has done or whether there are factors that cause the lawyer to judge the client in a negative light. See Jacobs, supra note 4, at 361, 379.

¹⁴⁶ Student materials describing Habit Four suggest a variety of strategies including: turn the conversation back to the client's stated priority; seek greater detail about the client's priority; give the client a chance to explain her concerns in greater depth; ask for examples of critical encounters in the client's life that illustrate the problem area; explore one example in some depth; ask the client to describe in some detail what a solution would look like; and use the client's own words.

¹⁴⁷ This analysis is critical if we are to begin to work on some troubling trends that Jacobs has identified. See note 73.

¹⁴³ A classic technique that lawyers use to confirm accurate understanding is to ask the listener to translate back what the speaker said. See John Barkai, Teaching Negotiation and ADR: The Savvy Samurai Meets the Devil, 75 Nev. L. Rev 704, 737 (1996). Another method used to confirm accurate understanding is role-play. See Mary Marsh Zulack, Rediscovering Client Decision Making: The Impact of Role-playing, <u>1 Clin. L. Rev. 593 (1995)</u>. Zulack argues that role-playing is a powerful technique that "opens a window to important decision making material for individual clients....." <u>Id. at 615.</u>

encourages the student to create settings in which bias and stereotype are less likely to govern. Second, the Habit promotes reflection and change of perspectives with a goal of eliminating bias.¹⁴⁸

Like the proverbial straw that breaks the camel's back, Habit Five recognizes innumerable factors that interact with bias and stereotype to negatively influence an attorney-client interaction. A lawyer who proactively addresses some of these factors may prevent a problematic interaction from reaching the breaking point. ¹⁴⁹ In giving the students the vignette of the asylum lawyer, we hope to communicate that this lawyer, like all others, brings biases and stereotypes to the attorney-client interaction but could take certain steps to lessen the impact of these biases. For most of us, when experiencing the anger that the asylum lawyer feels, the hard part is recognizing that our reaction to the client stems from bias rather than justified anger. Habit [*78] Five ¹⁵⁰ asks the student to acknowledge his every thought, including the ugly ones, and find a way to investigate and control for those factors that influence lawyering in unacceptable ways.

Research indicates that we are more likely to fall prey to stereotype when we are feeling stress and unable to monitor ourselves for bias. Thus, Habit Five suggests that students take simple proactive steps to lessen stress, including the type of stress the lawyer in the above example probably feels - taking a break, having food and drink and identifying what is interfering with his interaction with the client before the interview resumes. By taking these simple steps, the student will gain a greater capacity to monitor and hopefully avoid a biased reaction.

Habit Five thinking asks the student to engage in self-analysis rather than self-judgment. By engaging in this reflective process, the lawyer is more likely to respond to and respect the individual client.

IV. Teaching The Habits To Develop Cross-CulturalCompetence

In planning a cross-cultural class, a teacher should strive to develop awareness, knowledge, skill and motivation for learning in her students. The Habits are a way to gain greater knowledge and awareness as well as develop skills essential to cross-cultural lawyering. They raise our awareness by causing us to pay attention to the significance of differences and similarities and increase knowledge by gathering culture-specific information. The Habits also encourage the development of skills that are needed to gather and process information about clients and their culture. The selection of Habits on which a teacher focuses and the manner in which a teacher develops the concepts of cross-cultural lawyering are highly dependent on the teacher's background and skills and her students' background and skills. In this section, I describe teaching techniques that Jean and I and my clinical colleagues at CUNY have used to build cross-cultural competence by teaching the Habits. ¹⁵¹

A. Developing Motivation

¹⁴⁸ By tying this work to legal work and by showing the impact on lawyering, the other Habits hopefully encourage students to explore bias and stereotypes. Stereotypes and bias are significant barriers that limit the capacity of lawyers to establish relationships, to explore facts related to the individual, and discover suitable solutions for the client. See Earlene Baggett, Cross-Cultural Legal Counseling, 18 Creighton L. Rev. 1475 (1985).

¹⁴⁹ Habit Five derives from a theory of car accident prevention called final factor analysis. We've all heard of drunk drivers who miraculously made it home without killing themselves or anyone else, and also know that even the most careful drivers are not immune from car accidents. People reconstructing car accidents use final factor analysis to explain that a confluence of variables cause accidents when enough factors come together to form a critical mass. Thus, being drunk might not be enough in itself to cause an accident, but if the radio is blaring, the kids are fighting in the backseat, the driver is preoccupied with work, sleep deprived, and driving into the sun, one or more of these factors may combine with the driver's intoxicated state to cause an accident.

¹⁵⁰ Over time, we hope that students will learn to coordinate final factor analysis with Habit One to ensure the identification of factors about the client that may cause the lawyer to be more likely to stereotype and less likely to see the client within her own context.

¹⁵¹ Throughout this section of the article, I mostly talk about teaching ideas that "we" have tried. I have never taught these Habits alone, instead always having the benefit of the collaborative teaching that is part of clinical teaching at CUNY and also co-teaching with Jean at CUNY and at Yale.

Whether the teacher's goal is developing awareness of the impact [*79] of culture or skill in cross-cultural interactions, a lawyering focus will increase the likelihood of student receptivity to the material. ¹⁵² If the teacher uses primarily materials that contain examples of lawyering issues, students will be less likely to think that they are simply being asked to be a different (better) person. If the students see these skills as important to good lawyering, they will be more open to including them in their repertoire of skills. As I discussed earlier, resistance to this subject is to be expected ¹⁵³ and may arise for different reasons. Identifying the potential resistance and ways to address it are important aspects of planning classes and supervision.

On one occasion after teaching Habit One, I asked my students whether this Habit makes sense and I received two different and polar reactions. One student, a white feminist, female student in her late twenties, said, "This is CUNY, we do not really need to talk about differences and similarities. We do this kind of analysis all the time." The other student, a white male in his mid-twenties committed to becoming a public defender, commented that it was counter to the way he thought. He thought that by focusing on differences at all, we were denying the equality of all people. He did not think of people in terms of identity and culture, he said. Instead, he argued, all people are the same.

Both of these students were expressing a resistance to the teaching, but it was coming from different vantage points. The second student was articulating a view held by many students that, in a society with a history of discrimination and a current articulated commitment to equality, acknowledgment of difference violates that commitment. This student had little experience assessing the current impact of discrimination and lack of privilege of his clients in the criminal justice system. He also had little understanding of his clients' lives and how lack of understanding might impact his relationship with clients. Overcoming resistance of this sort requires teaching models that will allow the student to appreciate that equality is not inconsistent with difference and that will nourish the student's commitment to commonality [*80] with his clients. By explicitly addressing his concerns over time, we could hopefully help the student see that a failure to address issues of difference may in fact result in inequality or at the very least in misunderstanding. ¹⁵⁴

The first student may have been raising several different points of resistance in her comments. The first was that cross-cultural training was not necessary for progressive, well-meaning people. The teacher was, in a sense, "preaching to the choir." Or, secondly, if the student did need this information, the instruction was too elementary for her. If the material for instruction is only designed to develop awareness and is targeted to a specific group, some students may view the instruction as unnecessary. However, if the material is designed to focus on teaching multicultural analysis and skill and on helping each student identify their ethnocentrism, then all students have

¹⁵² Dark, supra note 90, at 16. Also, as Fran Quigley pointed out in his article on integrating adult learning theory in the teaching of social justice, adult learners find educational activities to be most meaningful when they are directly connected to students' concerns of becoming a lawyer, and if currently applicable, to representing a client. See Quigley, supra note 6, at 49.

¹⁵³ K. Cushner & R. Brislin, Key Concepts in the Field of Cross-Cultural Training: An Introduction, in Improving Intercultural Interactions 1-17 (K. Cusher & R. Brislin, eds. 1997). Brislin and Cushner begin their book by recognizing that intercultural interaction among human populations has typically been accompanied by violence and aggression. The authors note that even where people do not exhibit intense prejudices, interactions with those who are culturally different may provoke anxiety. Understanding this anxiety should help students understand the difficulties their clients face.

¹⁵⁴ The first part of the student materials with examples demonstrating the relevance of culture theory to lawyering were added in part to address the perspectives of students such as this second student. Clinical teachers who have heard a description of the Habits occasionally raise concerns about whether focusing on difference causes students to stereotype. Because Jean and I saw stereotyping occurring in lawyers and judges, we hope that a method that acknowledges it and consciously engages ourselves and students to compensate for our stereotypes would make the stereotyping less pernicious. See supra note 73 for examples of studies that show discriminatory treatment in the legal system. something to learn. Finally, the student's comment reveals an assumption of similarity that all of her classmates think like she does, which itself reveals possible learning opportunities. ¹⁵⁵

Although these exchanges signified resistance, neither student was hostile to the class. Hostility can be expected from some of the students, especially if they feel targeted and judged by the instruction. I recall early teaching experiences in which I tried to raise issues of racism and privilege and white male students felt targeted and reacted with hostility. ¹⁵⁶ These early classes accomplished few of the goals that I had for them. They did not help all students prepare for the racism they would face in the legal profession; they did not help students relate better to clients; and they reinforced what some students already understood about the world without really teaching skills for translating their knowledge.

[*81] Teaching the Habits and using a cross-cultural framework has worked better for the most part. In every class that we have taught, we have asked students for feedback. Mostly we get positive feedback from a broad range of students. Students of color and immigrant students generally say that they feel that their experiences are validated. Most students report that they have a better idea about how culture influences their lawyering. As teachers, we know that a single class on the Habits does not significantly influence the students' work at the skill level. However, with the vocabulary and framework taught in that class and in the materials, students are able to reflect in supervision on cross-cultural issues in their interaction with clients and the legal system and to learn from their experiences.

B. Raising Awareness Of the Significance of Culture

Introducing the Habits through classes and using supervision to encourage students to use the Habits will help students appreciate the significance of cultural similarities and difference for lawyering. Even after one class on the Habits, most students report an increased awareness of how culture might influence their lawyering. ¹⁵⁷ Teaching Habit One is a good starting place for raising awareness about the influence of culture because it asks students to articulate similarities and differences between lawyers and clients and begins a conversation about the effects of these factors on lawyering.

In class, we have used a variety of starting points to teach Habit One, including commencing with a focus on the cultural identity of the student or by asking students to apply Habit One to themselves and a student with whom they are paired or to themselves and their client. ¹⁵⁸ When students apply Habit One, we ask them to do both the Venn Diagrams and lists in less than five minutes. ¹⁵⁹

¹⁵⁷ We gave students a written feedback form that asked among other questions: "what's one thing you learned from this class?" The most common comment was a report of increased awareness of the potential effects of culture.

¹⁵⁸ I have never done all three together, but it occurs to me that, if I allocated the time for this exercise, the students would learn different things from each of the lists and diagrams.

¹⁵⁹ Sometimes we have started by first asking for Venn Diagrams - also known as circles - so that the student is able to get an overview of the relationship and the lists to give the student insight into her thinking about the client. For example, if a student shows very little overlap with the client in drawing circles but is able to come up with a long list of similarities, perhaps the student has learned the important lesson that similarities and differences are not equal. Perhaps the student sees the bottom

¹⁵⁵ In this class, after hearing these two reactions, I asked the students to assess the differences between these two positions and to consider what implications for lawyering might grow out of them. We discussed the differences for a few minutes and then I asked the students to hold their thoughts about these two positions until the end of class to assess whether there was anything new that they learned that would be helpful to their lawyering.

¹⁵⁶ Generally, this was the one class in the semester devoted exclusively to "diversity" issues. Visiting lawyers talked about racism, sexism and homophobia in the profession and its impact on clients. Professor O'Leary recalls a similar class focused on race in which an African-American student expressed outrage about the way the subject was presented and his fellow students' reactions to it. O'Leary, supra note 4.

In one class that we started by asking students to articulate their [*82] own identities, ¹⁶⁰ we got a list of what looked more like personality traits than cultural identities. Students answered the question asking for three characteristics that described themselves by responding with adjectives like "friendly," "assertive," "independent," "disciplined," etc. ¹⁶¹ After I wrote these on the board, I asked students to list three cultural characteristics, which I then also wrote on the board. Finally, I asked students to comment on the relationship between these lists. Was one list personal and the other list cultural, or were they connected?

We started with "friendly" and explored its cultural significance. The first student I asked to define "friendly" explained that she was from a small town in the Dominican Republic and that "friendly" means taking care of your neighbors. Because the government provided very few services in that region of the country, the student explained, people depended heavily on their friends for assistance - ranging from clearing hurricane debris to providing child care. When I asked the student who had listed "friendly" if that was what she meant by the term, the student explained that she was Italian and that, to her, "friendly" meant warm expressions of greeting, hugging, kissing, and so forth. ¹⁶² We explored what we could do to make sure we understand what a client means by "friendly." From there we moved on to whether characteristics like "assertive" or "independent" were related to culture and, if so, how. Students from more traditionally collective cultures explained that these traits were not as valued by their culture and that they would not normally list these traits even if they actually applied. Again, we asked how these differences might play out in the lawyer/client/legal system. ¹⁶³

[*83] In other classes, we asked students to apply Habit One to each other in pairs, listing all of the differences and similarities and then categorizing the differences and comparing the resulting roster with the list in the materials. In response to a "what did you learn?" question, one pair offered an example from their exchange. The first student listed economic differences because the other student was using an expensive pen. When the second student revealed that she had found the pen in the subway on her way to school that day, the students recognized the basic lesson that assumptions of similarity and difference can be inaccurate if based on unexplored facts.

After identifying differences and similarities between students, we ask students to apply Habit One to their relationship with a client. This application of the Habit serves to illustrate that we have different identities depending

line differently after the lists. By doing the exercise relatively quickly, the students will often list items that they hesitate to list if they were to devote substantial time to it.

¹⁶⁰ In her article on lawyering for native communities, Professor Zuni Cruz describes an exercise that is designed to focus the student on her own identity through use of a cultural chart outlining major cultural influences over time in the student's life. Zuni Cruz, supra note 4. Professor Hing describes a beginning exercise in his Lawyering Process class in which students interview one another to develop biographical sketches. Hing, supra note 4, at 1812. Professor Aiken notes that the differences among the students can be the best teacher of observing and valuing differences. Aiken, supra note 4, at 49.

¹⁶¹ Of course, these were not the characteristics I thought I would hear. My question did not specifically ask for cultural characteristics, but this was a class on cross-cultural lawyering and I assumed that the students therefore would focus on cultural characteristics. Perhaps, the students had not read the materials or perhaps they were resisting thinking about themselves in that way. As I scrambled to see whether there were notes in the margins of students' materials on their desks, I asked the more specific and direct question about culture.

¹⁶² Friendly" can be shown by polite and listening behavior; formal behavior; verbal and disclosing; physical and loud. See Pederson, supra note 107, at 114.

¹⁶³ No doubt when some of you read these stories you will note the cultural richness of the classroom exchanges at CUNY and at Yale. Jean and I are fortunate to teach in law school classrooms that are enriched by immigrants and children of immigrants, by diversity of sexual orientation, age, race, class and ethnicity as well as many other differences. Because we typically have a critical mass of students in these categories, students feel less like token representatives in the classroom. One significant way to enrich law school classrooms is working through admission and hiring committees to create a better learning environment for all students. See also Gary Orfield & Dean Whitla, Diversity and Legal Education: Student Experiences in Leading Law Schools (August 1999) at <http://www.law.harvard.edu/groups/civilrights/publications/lawsurvey.html> (reporting that in diverse educational environments 67-72% of students experience moderate or more enhancement in how they and others think about problems and solutions in classes and in learning and in their ability to work more effectively or get along better with members of other races).

on the individual with whom we are interacting and that we notice different things about ourselves when the reference point changes. ¹⁶⁴ Clients may also have different identities depending on the interaction and players.

In focusing on the client-student circles and lists, we often ask questions that are directed yet open-ended. Are there large or small overlaps? What are the relative lengths of the lists of similarities or differences? What's the connection between these two? We have asked students to compare the lists in the materials with the lists they have created and to consider whether there are differences. Students sometimes recognize that they have omitted something. For example, occasionally white students have noted that they omitted race as a difference when dealing with a client of color. ¹⁶⁵ Often, students have listed commonalties that are very specific to the case.

With open-ended discussion, we ask the students to begin to analyze the lists and diagrams and consider how these might affect their lawyering. Did they learn anything new or gain a useful insight into the case or client by making the lists? What might cause us to omit a **[*84]** characteristic like race that another lawyer or the client might list? Inevitably, we get a few good examples that foster awareness of the importance of culture to the attorney-client relationship. Since our ultimate goal is to build awareness, a wide range of examples can be used to convey the lessons the session is designed to teach. ¹⁶⁶

C. Developing Knowledge - Specific and General

One way to acquire general knowledge about the nature of culture is to identify the ways in which the law itself is a culture. Students learn essential aspects of culture theory, including the components of culture and the role it plays in determining meaning and shaping values and behavior. Students come to realize that law has its own language, ¹⁶⁷ values, ways of thinking, hierarchy, categories, rules and professional norms. ¹⁶⁸ The culture of argument and [*85] competition predominate and time is measured in precise terms.

¹⁶⁴ Mari J. Matsuda, When the First Quail Calls: Multiple Consciousness as Jurisprudential Method, 14 Women's Rts. L. Rep. 297, 298 (1992).

¹⁶⁵ Harrison & Montoya, supra note 4, at 410 (noting that part of white privilege is not having to notice one's race): Aiken, supra note 4, at 14 & n. 48 (noting the transparency of whiteness).

¹⁶⁶ Other teachers have used alternative methods for raising cultural awareness. Some have asked students to watch videos or listen to closing arguments while paired with people who are racially or culturally different and to compare notes or reactions. Professor Margelynn Armstrong presented the idea of watching movies and comparing notes at a SALT conference and Professor Martha Rayner described the closing argument exercise to me. Marty Peters describes using a scene from the movie "The Joy Luck Club" in which the story teller is explaining how her Caucasian American boyfriend misinterprets conversation by using literal rather than the subtle cultural meaning from the Chinese American context. Cochran et al., supra note 4, at 166. Other teachers have used books like Spirit Catches You and You Fall Down to raise awareness of how difference can create dilemmas for professionals in cross-cultural interactions. Anne Fadiman, Spirit Catches You and You Fall Down: A Hmong Child, Her American Doctors, and the Collision of Two Cultures (1997). This is a powerful and compelling story of a clash in the professional culture of medicine and the Hmong culture. Because the clash is so clearly identified by the author, it is easy to see it in a way that the people at the time did not. Professor Kathleen Sullivan assigned this to her first-year professional responsibility class at Yale.

¹⁶⁷ See Elizabeth Mertz, Teaching Lawyers The Language of Law: Legal and Anthropological Translations, <u>34 J. Marshall L.</u> <u>Rev. 91, 94-102 (2000)</u>. After studying contracts classes at eight law schools, Professor Mertz concluded that legal language and socratic dialogue are linked to the development of the capacity to "think like a lawyer." In the process of learning this new way of thinking, law students are taught to think in ways that decontextualize conflict and to view the conflict only within a legal framework. In the process, students are expected to abandon ethical contexts and replace them with relevant precedential categories. "Students thus begin to learn a process of translation that they will eventually take for granted." <u>Id. at 110</u>.

¹⁶⁸ Of course, the legal culture mirrors the majoritarian culture in many significant ways. See Richard K. Sherwin, When Law Goes Pop: The Vanishing Line Between Law and Pop Culture (2000). Despite law's connection to majoritarian culture, all law students experience some adjustment to the legal culture. Mertz, supra note 167. See Kimberle W. Crenshaw, Toward a Race Conscious Pedagogy in Legal Education (Foreword: Voting Rights: Strategies for Legal and Community Action), 11 Nat'l Black L.J. 1, 3 (1989) on reactions of students to color to the neutrality of law school teaching that masks the values and perspectives reflected in law. "Few professionals fully understand what a citizen experiences; the very process of becoming a professional

We have asked students to identify examples of the changes in thinking, communicating, and valuing that they have experienced over the course of two years of law school. Even if students did not themselves detect transformations of the content and style of their communications, undoubtedly friends and family did, and we ask students to talk about those reactions. ¹⁶⁹ This discussion facilitates an exploration of students' reactions to the legal culture and the existence of any culture clashes.

We ask students to think about the stress they feel upon entering a courtroom to represent clients and to connect this experience to the stress of entering a culture in which the cultural rules are hidden and one does not know how to predict the responses of others. ¹⁷⁰ For example, the introductory rituals of the courtroom are not clear and comfortable for the students. Because the legal culture is not quite as hidden to students as the cultures in which they have been raised, exploring the legal culture presents an opportunity for them to learn about culture generally.

Other ways to teach the general topic of culture is to look at the chart of cross-cultural themes in the materials and to ask students to identify how the particular theme might play out in law practice. By telling a story about their own client interaction or other lawyering they have observed, or by using examples from the material, students can explore cultural themes and their relationship to lawyering.¹⁷¹

Both Jean and I work in clinics that have clients from many different cultural groups. Accordingly, we are less likely to focus on a particular culture as part of our cross-cultural lawyering classes. However, if the clinic works with clients from a particular culture, the teacher will want to introduce the students to that culture. Fiction, [*86] movies, history, geography, food, customs and language are all roads into the client's culture. ¹⁷² If the clinic works with clients from different cultures, each student should be encouraged to learn about his or her client's specific culture and background. I have found that recent immigrants particularly appreciate a demonstrated interest in and knowledge about their country of origin. ¹⁷³

Conversations with clients are often a good way to develop or test culture-specific information. In simulated interviews and planning for client interviews, students should be encouraged to identify the ways they have or will

¹⁷⁰ A connection can be made between the stress the students feel upon entering the courtroom and the stress their clients must feel. While the legal culture of the courtroom is not fully known to the student, it is truly foreign to most clients.

¹⁷¹ If the stories in the materials are edited to include some of your own, the students will appreciate how cultural themes arise in the context of the cases on they are working. In addition, teachers can design critical incidents that highlight a particular theme. For example, if we want to teach the cultural theme of time, we might show the students an interview in which the lawyer is trying to build a time line as recommended in Binder & Bergman, supra note 36, with a client for whom time is measured very differently and ask the students what is happening. Or we might show how different cultures use silence by presenting an interview in which the client was silent to a degree that would make most lawyers uncomfortable.

¹⁷² Many other professions have materials focused on particular groups with chapters on different ethnic groups and suggestions about how to interact with members of the particular group. See, e.g., Freddy Paniagua, Assessing and Treating Culturally Diverse Clients (1998); Elizabeth Randall-David, Strategies for Working With Culturally Diverse Communities and Clients (1989); Derald Sue, Counseling the Culturally Different (1981).

¹⁷³ The Internet makes this kind of information easily accessible. Students can look at maps of their client's home country, read U.S. State department literature on the country and find relevant stories in international newspapers.

often extracts from us some measure of ability to relate to the perspective of people without this training." Margulies, supra note 88, at 727 n.160.

¹⁶⁹ To explore the impact of legal culture, Professor Hing uses a case study of the redress movement for Japanese Americans who were interned in World War II. He compares lawyers' and organizers' differing views of the possibility of success and questions whether legal training changes the individual's culture and class such that he or she is no longer the same as the ethnic or racial community of origin. Hing, supra note 4, at 1821.

gather culture-specific information. ¹⁷⁴ A series of questions that can be used for this purpose are described in Habit Four in the materials. ¹⁷⁵

Because culture is very complex, students should be encouraged to gingerly use the new knowledge they are acquiring. ¹⁷⁶ When we [*87] learn "specific cultural rules," we have to be careful to apply them correctly and to guard against substituting them for information about the client. Significantly, the client may or may not follow the "rules." Thus, the student who has worked with two or three clients from a particular group should scrupulously avoid generalizing from this experience to conclude that all people from this group are (you can fill in the blank).

What is the proper use of the student's knowledge about the specific cultures and about culture-general themes? The student can use this new information to develop different hypotheses about client behavior based on greater cultural awareness and can explore these with the clients. ¹⁷⁷ The more the student knows, the more the student's antennae to difference and similarity will be raised. Being attuned in this way is a critical cross-cultural skill.

Students who work with clients from a particular client group should also be encouraged to learn the client's language, if different from their own, especially if they plan on practicing law with this client group. ¹⁷⁸ Culture and language are connected to each other and many concepts are hard to translate because they have no direct corollary - and therefore no specific term - in the other culture.

¹⁷⁴ Peter Margulies suggests that we use narrative and counter narratives to create opportunities to engage one another's perspectives. Margulies, supra note 88.

¹⁷⁵ Anne Fadiman gives a list of questions developed by Arthur Kleinmen, the Dean of the Department of Social Medicine at the Harvard Medical School to address cultural differences. Fadiman, supra note 166. Fadiman uses this set of questions to show the contrast between the Doctor's assessment that the child in the book is suffering from Epilepsy and the parents' assessment. She hypothesizes how the parents might have answered the questions if they had been asked:

"What do you call the problem? Qaug dah peg. That Means the Spirit Grabs You and You Fall Down.

What do you think caused the problem? Soul Loss.

Why do you think it started when it did? Lia's sister, Yer, slammed the door and Lia's soul was frightened out of her body."

"<elip>What is the most important result you hope she receives from this treatment?" "<elip>We hope Lia will be healthy, but we are not sure we want her to stop shaking forever because it makes her noble in our culture, and when she grows up she might become a shaman."

Id. at 260.

Other questions included: What are the chief problems of the illness? What do you fear most? What do you think the illness does? How does it work? With these questions, the doctor is able to elicit culturally-relevant information about the nature of the illness and treatment.

¹⁷⁶ When I have read these chapters in the cross-cultural books, I find myself feeling wary of using the information. As an outsider, I have no way to evaluate the information. I do not know what changes have taken place in the culture since the chapters were written. They often seem a little too sweeping or general and not necessarily applicable to the individual. On the other hand, the chapters often present different ways of looking at a problem and can be used to create awareness of different approaches. The key to proper use of this material is to make sure that we do not apply it unconsciously. All of these books start with caveats that each patient, client, student is different and may not reflect the culture being described in the chapter.

¹⁷⁷ Jacobs, supra note 4, at 401, 409 n.294, citing Sue et al., supra note 57.

¹⁷⁸ There is a variety of language-related issues that should be considered. First, a shared language will result in a better relationship with clients. In fact a capacity to communicate in the client's language may be more important than ethnicity for forming a connection with clients. Hing, supra note 4, at 1818. In addition, language can shape the story told by clients. For example, native French speakers who are bilingual in English will tell a far more romantic and emotional narrative in French than when telling the same story in English. Students in Hong Kong express more traditional Chinese values when taking a values test in Cantonese. Even if students speak the client's language, they do not know how to express the legal concepts. We have offered a class at the law school in Spanish for Lawyers for Spanish speakers to learn how to express in Spanish the legal concepts that they learned in English. I did not always appreciate the difficulty for students who had to not only think about the legal concept in "plain English" but also "plainly" in some other language and culture linking it to some concepts that the client

D. Developing Cross-Cultural Skills

The Habits are designed to help develop analytical and interaction skills. Many of these skills have been taught in the context of other legal work. By furnishing a cross-cultural context, we enhance the students' capacity to employ the skills by focusing attention on why the skills are needed and how they might be used to address potential [*88] problems in client interaction.

1. Raising Antennae To Recognize Ethnocentric Thinking

a. Habit One: Similarities and Differences

The evolution from being able to talk the talk of "cross-cultural" to walking the walk of "cross-cultural" requires a capacity to recognize when we are making assumptions and judgments about our clients that grow out of our own cultural blinders. If students can attain this heightened awareness, they can begin to develop competence. Habits One and Two encourage students to focus on difference and similarity and their effects on lawyering. By using these habits, students will begin asking themselves questions such as: Why am I judging this client negatively? Is it because we have different values, experiences, or opportunities? Sometimes, as teachers, we have difficulty raising these questions. ¹⁷⁹ Students can also use Habit One to reflect on interviews and identify areas that are left unexplored because of assumed similarities. ¹⁸⁰

b. Habit Two: The Three Rings

Habit Two develops skills of analyzing the effects of similarities and differences among the client, legal system and the lawyer. Because of the complexity of these factors and their interactions, Habit Two needs a detailed concrete illustration to have it come alive for students. To teach Habit Two, we have used a simulation developed by Jean for her Children and Families clinic. Teachers should tailor these materials to their particular setting, using a simulation or case to develop the circles and lists of Habit Two. In each subject area and local region, there are cultural norms and values that will be specific to the law and legal system in which the students are working.

Because students need the practice of applying the framework of culture to analyze a case, Habit Two is one that we also often teach and reinforce in supervision. We have found it especially useful when we or the students are troubled by a case. Often, it helps the lawyer move away from irrelevant facts or issues about the client or the case that are bothering the lawyer and to focus instead on facts and issues [*89] that will be relevant to the fact finder.

Habit Two analysis can help the student spot weaknesses and brainstorm possible ways to explain the client's behavior or statements to a court that may be unfamiliar with the client's culture. The student may identify similarities between the court and client and build an explicit connection to the fact finder with these similarities.¹⁸¹

understands. Students can prepare for this complication by thinking about how they will describe the concepts to the client in their non-legal language.

¹⁷⁹ When I first started teaching collaboration skills, I noticed that the conversation with students about issues they were having with their co-counsel shifted from blaming comments about how their co-counsel was "screwing up" to brainstorming and questioning why difficulties were occurring. See Bryant, supra note 16.

¹⁸⁰ Professor Hing describes a video involving interviews and other legal work in which students explore the effects of similarities and differences between lawyer and client. Using their cases in the clinic, students discuss whether a lawyer who shares a cultural heritage with a client has an advantage and what students who lack such a commonality of culture can do to develop a trusting relationship. Hing, supra note 4.

¹⁸¹ If we recognize that seeing similarity causes us to be less judgmental about our clients, we need to think creatively about how to make connections between clients and fact finders so that they too can be less judgmental. See Marc A. Fajer, Can Two Real Men Eat Quiche Together? Storytelling, Gender-Role Stereotypes, and Legal Protection For Lesbians and Gay Men, <u>46</u> <u>U. Miami L. Rev. 512 (1992)</u>, for a thoughtful analysis of the benefits of identifying similarities to correct incorrect "pre-understandings," and the limitations of not recognizing difference, especially the difference that grows out of the effects of discrimination. <u>Id. at 528.</u> If the client is resisting the attorney's advice, the lawyer should try to identify the reasons for the client's behavior. Is there something in the client's background or culture that causes this resistance? The student should consider whether she is asking the client to do something that violates the client's norms and, if so, whether there are other possible approaches that might resolve the case. As Kimberly O'Leary recognizes, the identification of difference is a starting point for creative legal thinking; ¹⁸² so too is the recognition of similarities between client and court and lawyer.

Like the conversations suggested by Habit One, a regular dialogue between supervisor and student on issues of difference and similarity can make the conversations less loaded because they are not limited to occasions when the student has done or said something that is ethnocentric.

c. Habit Five: The Camel's Back

In classroom teaching of Habit One, a teacher can also introduce Habit Five. Habit Five depends on the analysis of similarities and differences from Habit One and on recognizing some of the more pernicious effects of bias and stereotyped thinking. Once the framework for Habit Five is laid in the classroom, Habit Five conversations are more insightful for students when they are applied to the student's fieldwork. This exploration might be accomplished by discussing whether the student's behavior or thinking would be different if the characteristics of the client changed. ¹⁸³ The challenge here is to strike **[*90]** the right balance of support and challenge in the conversation with the student. The goal is to provide the students with sufficient support and information ¹⁸⁴ so that they can challenge themselves. One way to do so is to acknowledge our own assumptions and biases when they occur.

- 2. Remaining Flexible and Making Isomorphic Attributions
- a. Habit Three: Parallel Universes

Teaching Habit Three is easy. ¹⁸⁵ Students grasp the concept readily and its use reinforces the critical skill of remaining flexible and making isomorphic attributions. We usually introduce the Habit by connecting it to the concept of attribution, explaining that cross-cultural interactions often result in lawyer and client ascribing different meaning to the same behaviors. Habit Three invites students to imagine alternative explanations for the client's behavior.

We have taught Habit Three by using videotape and the students' cases. One of our video scenarios is the one transcribed at the beginning of this article. We show this short passage on tape and ask, "Why is the brother here? n186" We ask students to ponder this question a^d then to brainstorm possibilities other than the one that initially occurred to them. Then, we list all the different explanations of the behavior on the board. Some possible explanations include that he is the driver; interpreter for language and/or culture; supporter; fellow tenant, in possession of relevant facts; the censor, to make sure she does not say something she shouldn't; an aid in

¹⁸⁴ As Professors Jacobs and Aiken point out, students often fail to see privilege and bias, and the challenge for us as teachers is to give the students experiences and knowledge to allow them to appreciate the effects of bias on their own thinking. See Jacobs, supra note 4, and Aiken, supra note 4.

¹⁸⁵ Applying it on a regular basis is of course a different matter. If students begin to make it a Habit, they will usually say before I have to - "I guess I should do some parallel thinking here."

^d Jean and I demonstrated this part of the class at the clinical teachers' conference in Albuquerque, in May 2000. This tape segment can also be used to teach Habit One.

¹⁸² O'Leary, supra note 4.

¹⁸³ Professor Gunning uses a dispute between a child and a bus driver to portray one possible method of assessing how cultural myths would change if the race, age or ethnicity of the participants changed. This technique, as illustrated in Professor Gunning's article, is a way of exploring cultural myths without owning them and thus, allowing the student to acknowledge stereotyped thinking without subjecting themselves to shame. Gunning, supra note 4.

decisionmaking or the intended decisionmaker (by her or his choice); and so forth. After we identify the various possibilities, we ask how our answer to that question influences our lawyering choices.

We then return to the tape and show a segment in which the lawyer explains to the brother that she is going to speak to his sister alone.

[*91] Attorney: Maybe what we'll do if it is OK with you is we will start the interview in the office just with your sister and then if we need some information from you we'll call you in afterwards. Is that OK?

Translator: [Spanish: The translator translates this to the client asking the client (not the brother) if it is OK with her.]

Client: [Spanish]

Translator: If you think it will be more convenient for her to do, but this is the person that always helps her.

Attorney: Do you prefer for him to be in with us?

Translator: [Spanish]

Client: [Spanish]

Translator: If you think it is going to help her.

Attorney: Well, why don't we start just with you and then if it turns out that we need him, we'll have your brother come in later.

We ask the students to look at the long list of explanations for the brother's presence and to connect the lawyer's actions to the list. Given what the lawyer did, what were her assumptions about the brother's reason for being there? We ask the students to identify other choices that the lawyer made in structuring the interview. We also ask the students to identify the potential benefits and costs of inviting the brother into the interview room.

Students typically identify a number of possible benefits of the brother's coming into the interview room, including providing support for the client, furnishing additional facts and language and cultural translation. We discuss the possible advantages of using him as a translator rather than the office translator. Students also usually identify several possible problems, including confidentiality issues; that his voice may dominate; he may translate and change the meaning; or he may silence the client. Students often have definitive positions about whether the sister is more likely to give the lawyer information if her brother is with her than if she is alone. Often these positions are tied to their own experiences and cultural background. For example, students from more collective cultures and immigrant cultures often explain that the brother is an essential part of the way in which the client interacts with the English-speaking world. Women students with strong feminist perspectives argue that the client should be interviewed alone to ensure that it is her voice that the lawyer hears.

Parallel universe thinking allows students to see that there are many possible interpretations of the brother's and sister's behavior. By interpreting the behavior and responding to it, students can begin to appreciate that they are substituting their own cultural explanations [*92] and norms for the behavior and are making lawyering choices based on these explanations. ¹⁸⁷ We usually end with a suggestion that parallel universe thinking alerts us to the possibility that we need additional information to develop a culturally appropriate response. ¹⁸⁸

¹⁸⁷ This scenario also provides an opportunity to teach the legal norms as culture. Most confidentiality rules are based on a highly individualistic norm.

¹⁸⁸ Exactly how we get this information may be tricky. Directly asking the client will not work if she does not feel free to say she wants her brother excluded, but additional information will help the lawyer assess this issue.

Including a cautionary note in teaching Habit One and the other Habits allows a teacher to identify the limitations of lists and charts and of culture-centered training. In addition to understanding that we are constantly changing identities, we also need to acknowledge that we may move quickly to sort and order to lessen ambiguity. This may result in our ignoring the complexity of the situation and the relationship, and perhaps contributing to a different kind of misunderstanding. Thus, such situations require flexibility and the ability to be comfortable with not knowing or changing the lists and diagrams to incorporate more and different information.

3. Remaining Non-Judgmental

In our classes, we explicitly address the importance of remaining non-judgmental towards clients and ourselves as learners. We talk about the highly judgmental nature of the legal culture and the special difficulty that lawyers may experience in cross-cultural encounters as a result of our training. To teach the skill of remaining non-judgmental, we use Habits Three, One and Five.

a. Habit Three

In one class, we asked the students to recall a negative judgment that they made about a client's behavior. We asked them to write down the behavior and their interpretation of the behavior. Then, we asked them to use parallel universe thinking to identify at least three other possible explanations of the behavior. When we asked for volunteers, one student identified an instance in which she assumed that her client was no longer interested in a housing eviction case because the client did not respond to the student's phone call and letter. We wrote the behavior on the board and asked the student and her classmates for alternative explanations. We stopped after approximately twenty different interpretations had surfaced. Then, we asked about the sources of the interpretations and how they connected to our experiences and culture. ¹⁸⁹

[*93] This technique of exploring alternative explanations is commonly used in clinical teaching to expose students to the limitations of relying on their own experiences to interpret client behavior. ¹⁹⁰ By teaching the technique as a cross-cultural lawyering skill, students learn the importance of searching for alternative explanations. ¹⁹¹ They learn to hold off making judgments until they have considered what additional information may be needed to make sense of the behavior. ¹⁹²

b. Habit One

¹⁸⁹ This conversation, which occurred in Jean's Families and Children Clinic class, was enriched by the experiences of her students and was an example of how classroom conversation can sometimes be richer than discussion in supervision because of the involvement of the entire class. As might be expected, students who had experienced loss of housing themselves as well as students who had worked in prior careers as social workers and community organizers involved with public housing tenants and poor people were able to add perspectives based on their experiences.

¹⁹⁰ See Quigley, supra note 6, at 57; Lois Johnson & Louise Trubek, Developing A Poverty Law Course: A Case Study, 42 Wash. U. LJ. Urb. & Contemp. L. 185, 199 (1992) and Aiken, supra note 4, at 29 for other examples of classroom conversations that led to students identifying multiple explanations for client behavior. Mary Zulack and Conrad Johnson at Columbia conduct an exercise at the beginning of their clinic program that involves a similar scenario in which they ask the students to come up with multiple reasons for the client behavior. This exercise is described in detail in Aiken, supra note 4, at 28.

¹⁹¹ This is an example of an instance in which teaching the theory behind the skill enhances the likelihood that the skill will be learned and used. See supra notes 57-76 and accompanying text.

¹⁹² This is a good example of how a personality trait - tendency not to judge - that is valuable to cross-cultural interactions can be developed as a skill. By teaching students the questions to ask and methods for analyzing situations, we teach the skill of suspending judgment. Professor Jacobs recommends that before students inaccurately classify clients as difficult, students should ask themselves: "Is there something I said, did or thought which prevented the interview from concluding more positively? Can I detect any factors that have motivated me to evaluate my client in a negative light? Are there gaps in my understanding of my client's experiences and world view that prevent me from fully understanding what my client's needs are?" Jacobs, supra note 4, at 361.

In addition to Habit Three reasoning, students can learn to use Habit One thinking when they are feeling judgmental towards a client or client's behavior. Searching for similarities, an integral part of Habit One, is another useful way to counter negative judgments about clients.¹⁹³

c. Habit Five

An essential component of Habit Five is to address one's bias and [*94] stereotyping in a non-judgmental way. In teaching the Habits, a teacher needs to set a similarly non-judgmental tone in the classroom. I often start with canned video, and if I am not careful, I can inadvertently create a judgmental atmosphere by critiquing videotapes harshly. I am much less likely to pay attention to positive aspects of a performance or to framing criticism in a positive manner when the performers on the tape are not students in the class. Yet, students often see themselves in the tape and respond to the criticism as if it were them. ¹⁹⁴ Thus, it is important to build an atmosphere of support and challenge that includes the person on the videotape as well.

4. Cross-Cultural Communication Skills

The practice of Habit Four increases students' skills at cross-cultural communication. We have taught Habit Four primarily through watching and interpreting videotapes, conducting simulated role plays in class and engaging in reflection in supervision of meetings with clients. These methods of teaching mirror the ones I use to teach interviewing skills. I design short exercises, simulations or videos to focus on parts of an interview for demonstration and analysis. For example, focusing on the beginning of the interview, we can identify the different introductory rituals that cultures use and we can role play the integration of these into the attorney-client interview.

To help students acquire the deep listening skills that cross-cultural interaction requires, we expose students to exercises that help the student become mindful of the interviewing process. ¹⁹⁵ Deep listening is difficult for students to master because many western cultures undervalue listening. Most students who were encouraged in their childhood to pursue a legal career probably received this advice because they displayed a tendency to argue, not because they were good listeners.

In watching videotapes or reflecting on case interviewing, we ask whether we are correctly interpreting the client and she us. Especially when we are using translators or working cross-culturally, the lawyer must prepare for the possibility of some misunderstanding. We ask questions designed to surface the cues the students are using to understand the client. These include: What evidence do we have for our interpretation? What non-verbal cues are we using and is our interpretation [*95] correct? ¹⁹⁶ Is the client's style different from our own and can we separate

¹⁹⁴ One student gave this feedback to us following a conversation about a tape that we had used in class. See supra notes 86-102 and accompanying text for a discussion of the importance of establishing an appropriate mixture of support and challenge.

¹⁹⁵ Professors Harrison and Montoya have suggested a process similar to deep listening skills, which they call "slow-motion" reading, to capture a similar idea that we need to listen on many different levels and to make explicit the multiple layers of communication. See Harrison & Montoya, supra note 4, at 433.

¹⁹⁶ In the counseling field, trainers recommend noting and using the client's preference of eye contact and paying attention to the client's body language as well as testing whether the counselor's body language and verbal language are consistent with the client's. See Pederson, supra note 107, at 105. Lawyers interviewing clients can follow these same tips to establish rapport with clients.

¹⁹³ At a faculty forum at CUNY, I presented my work by asking the faculty to identify a student about whom they were feeling very negative and to apply Habit One to the student and themselves, identifying similarities and differences. In our conversations about what they noticed as a result of the lists and circles, teachers talked about searching for similarities and how the exercise had made them think differently about their relationship with the student. Some said they needed to know more about the student because they could only list differences; others said it helped them focus on what it was about the student that caused them to have the negative reaction.

content from style when the style is so different from our own?¹⁹⁷ What other steps could we take to confirm that our interpretation is the correct one?¹⁹⁸ How many times did we change the topic and what was the result?¹⁹⁹

Learning deep listening skills proves difficult for students because they are trying to put together many different tasks and sub-tasks in client interactions. Thus, one key piece of the deep listening of Habit Four is to alert students to look for red flags - clues that something is going wrong in the interview. By teaching students this, we implicitly acknowledge that these interviews are difficult and that we often can make mistakes. Recognizing and responding to red flags, which is a complicated task in any interview, requires even greater attention and skill in a cross-cultural interview. Students can be taught to look for red flags by reviewing simulated interviews and by making flag notations in their client interview notes. ²⁰⁰

In talking with students about their red flag analyses of tapes of cross-cultural interviews, the teacher can encourage the kind of reflection that Professor Jacobs suggests for effective client-centered lawyering. For example, Professor Jacobs encourages students to reflect on their role in the interview and the non-verbal cues they may be sending if they find themselves labeling the client as difficult or uncooperative.²⁰¹ [*96] Reflection on the means available to improve the communication may also allow the student to consider how their clients' experiences with other lawyers or others from the lawyer's cultural group may complicate the interviewing process.

As part of developing the skill of debriefing an interview, students should be asked to examine what they followed up on and what they left unexplored. To teach the connection between Habit One and communication skills, I have used two different videotape segments from the same videotape we use for Habit Three. One segment involves a non-payment of rent issue:

The landlord is saying that she doesn't pay her rent, but she does pay her rent. She doesn't know why he is saying that.

Attorney:Who do you live with in your house? Client: [Spanish] Translator:She lives with her brother, her husband and her two children. Attorney: Whose name is on the lease? Translator:[Spanish] Client: [Spanish] Translator: Her husband and her name. Attorney: Who's the person that usually pays the rent? Translator: [Spanish]

¹⁹⁷ Determining meaning from clients' tone and speech rate is complicated especially when the counselor or client is not using a native language or translation is used. Pederson, supra note 107, at 106. If possible, clients should be allowed to communicate in their language of choice to mitigate some of these problems.

¹⁹⁸ Some of the suggestions we use include asking the client to tell us what we have told her; using active listening techniques to replay for the client what she has told us; and engaging in role plays of the sort suggested by Professor Mary Zulack in Rediscovering Client Decision-making: The Impact of Role Playing, supra note 143, at 618. Perhaps one of the reasons for the efficiency of the role play approach is that clients with an indirect communication style may be more forthcoming in role than if questioned.

¹⁹⁹ In a text prepared for teaching culture-centered counseling, Professor Paul Pederson recommends staying on the client's topic as long as possible as a way of seeing the client's problem within a cultural context. Pederson, supra note 107, at 106. A similar approach makes sense for lawyers. Facts and stories may not make sense to the lawyer, but it is important for the lawyer to understand why and how they make sense to the client.

²⁰⁰ One of the difficulties in teaching students this skill of deep listening is that it is a skill that we must observe the student employ if we are to give the student feedback. If the bulk of the student's interviews are not observed or videotaped, feedback on the non-verbal communication and interpretation of client behavior is difficult to provide. Therefore, a teacher interested in teaching her students this skill has to provide multiple opportunities for observation and feedback.

²⁰¹ Jacobs, supra note 4, at 17.

Client: [Spanish] Translator: Who writes the check? Or who is the one who gives the money? Attorney:Both. Translator: [Spanish] Client: [Spanish] Translator: Her husband works, but she is the one who writes the check out. Attorney: Is that how you pay the rent, by check? Translator: [Spanish] Client: Si. Translator: Yes. Attorney: How do you usually give the check to the landlord? Do you hand it to him? Do you mail it in? Translator: [Spanish] Client: [Spanish] Translator: She sends it by mail. [*97] Attorney: I imagine that you did not bring them today, but you have your cancelled checks from the other times that you paid? Translator: [Spanish] Client: [Spanish] Translator: Yes, she has them. Attorney: From what you are saying, you have been paying your rent? Translator: [Spanish] Client: [Spanish] Translator: Yes, she pays it. Attorney: Is there a date when you have to pay it by every month? Translator: [Spanish] Client: [Spanish] Translator: She usually pays before the end of the month. Attorney: Before the end of the month. Did the landlord ever tell you the day that he wants the check by? Translator: [Spanish] Client: [Spanish] Translator: Sometimes she pays at the beginning of the month. Sometimes at the end of the month. There are 30 days to pay the rent. She pays between those 30 days. Attorney: Have you ever had to pay your rent late?

We use this segment to show how difference may influence questioning. I ask the students why the lawyer is exploring time and payment. Presumably, something does not make sense to the lawyer. The lawyer knows that a defense to non-payment will require a showing of payment of the rent. The lawyer is exploring when rent is paid and continues exploring this subject because the story strikes the lawyer as odd. A lease would have a date, but it seems as if the client does not pay on a certain date. Time is a culturally bound concept that may not have the same meaning for lawyer and client. What does "late" mean? Legal definition? Cultural? How do the lawyer and client view the landlord-tenant relationship?

This segment also shows assumptions of similarity. Asking "whose name is on the lease" rather than "do you have a lease" or "who pays the rent" reveals assumptions of similarity by the lawyer. Fortunately, the client saves the lawyer from one of the ambiguous questions by asking the lawyer what does she mean by pay. In one class, a student pointed to the question about who pays the rent as one that revealed a cultural assumption. In the student's immigrant family, bills were placed in a bowl at the entrance to the home and were [*98] paid by whoever had enough money to pay the bill that month. Generally, a different person paid the rent each month.

To acquire culturally diverse conversation management skills, students need to learn to be less rigid about the structuring of an interview. The issue of who is supposed to talk and for how long and about what are very much related to culture. Direct and indirect acquisition of information may occur and students who remain flexible and listen deeply will achieve better client interactions.

The other segment that I use to teach the connections between questioning and culture is one in which the lawyer learns that domestic violence may be a factor in the client's landlord-tenant issue:

Attorney: Would you like to talk a little bit more about different issues around the immigration situation of your husband and perhaps about the battering situation? Do you think that that might be helpful for you? Translator: [Spanish] Client: [Spanish] Translator: If you want to continue to talk about it. She has been speaking to her pastor at church and he has been giving her advice. Attorney: This office, this legal services office, we do have some people who specialize in issues of immigration and also in problems in the home, in domestic violence problems. Translator: [Spanish] Client: [Spanish] Client: [Spanish] Translator: My sister was here before and she was helped. That's why she came. Attorney:We can begin to perhaps talk about some of these different issues. Translator: [Spanish] Attorney: But that does not mean that ... you should definitely keep talking with your pastor and if your brother is a good person to talk to - continue getting support from other places as well.

Translator: [Spanish]

The lawyer does not ask questions about the pastor and whether the client has found the pastor's advice useful or supportive. Most likely, the lawyer assumes that the client has a relationship with the pastor that mirrors such a relationship in the lawyer's life and thus, does not pursue the questioning because the client's actions make sense. Contrast for example the presumption that the pastor is helpful [*99] but that the brother might not be. Perhaps the lawyer simply assumes that conversations with the pastor are private and not to be discussed. In addition to illustrating the connection between assumption of similarity and failure to question, the lawyer's behavior can be used to illustrate parallel universe thinking by asking students to imagine all the reasons why the lawyer responded in the way she did.

Conclusion

If we can make the subject of cross-cultural lawyering one that our students and we can talk about, our collective capacity to practice law in non-discriminatory and culturally-sensitive ways will increase access and substantive justice for our clients. By developing a framework and vocabulary for cross-cultural lawyering, Jean and I hope to increase our students' and our own capacities to talk and think about difference and similarities in a more complex manner. If we lessen the anxiety of interacting with others whom we perceive as different, we promote an openness to learning more about our clients and ourselves.

By thinking about this work as creating Habits, we enable students and ourselves to explore issues of difference in a routine way that, we hope, makes conversations less loaded. This spares the teacher the difficulty of figuring out how and when to raise issues like race or gender, stereotyping, bias or others. Instead, the conversation takes place on a regular basis within the framework of the Habits.

Our work on the Habits and learning about cross-cultural teaching in other fields has been among the most stimulating work I have done as a clinical teacher. There are many good resources out there from which we can learn, and we need to continue to create paths to greater cultural competence for ourselves and our students. I agree with Professors Montoya and Harrison that

cross-cultural learning is possible only in tandem with others. It is a journey that cannot be taken alone. The borderlands are places of collaboration, of interactivity, of shared as well as opposing values, of exposed and juxtaposed weakness, and of ignorance, unmasked and remasked. Borderlands beckon to risk takers, meaning awakers, and vision makers.²⁰²

Thus, I end where I began: by thanking colleagues, staff, students and Jean, who have been learners and teachers and who have made this project an enriching journey. As clinical teachers, we have an enormous contribution to

²⁰² Harrison & Montoya, supra note 4, at 441.

make to our profession and the clients it serves if we can build cross-cultural competence in our students and in [*100] ourselves. ²⁰³ By using the multi-cultural resources and starting the conversations, we start or continue an important journey for ourselves and our students.

Epilogue

by Sue Bryant and Jean Koh Peters When Jon Dubin, Sue's editor for this article, asked us to write about our collaborative experience in an Epilogue, we said yes because the process of working together to develop the habits of cross-cultural lawyering was fun and enriching for both of us. We wanted to record, for the benefit of our colleagues, some of our ideas about long-distance collaboration on a project. Although we used a process geared to our particular work styles and learning styles, we hope that it sparks some ideas for collaborations by others.

I. The Process

A. Collaborating in Conversation

Our core process was to converse regularly to yield the classes, the class materials and writing that have become the Habits of cross-cultural lawyering. From the day we began, we started with a weekly hour-long phone conversation that continued throughout the years of the project. Typically, one of us, usually the one who was in the office as opposed to working at home, called the other, at a prearranged time that we had set aside.

We scheduled phone conversations (which we usually called meetings) into our weekly semester-long schedules, trying to be mindful of the need to work around traditionally busy court time. Because the meetings were regular, we were flexible about starting late when needed or canceling when needed due to the inevitable clinic case emergencies. But if a meeting was canceled, we tried whenever possible to reschedule it for the same week.

The conversations themselves were very free flowing. We often shared conversations that we had with students in supervision about their cases and our own observations about the ways in which cross-cultural issues were arising in our work. We shared stories of our own cultural blunders and of applications of the Habits that seemed useful in helping students deal with issues. We moved quickly back and forth between clients, students and our families as we tried to look for examples [*101] in our lives that confirmed or challenged what we were reading and thinking. Because we are good friends - and became better friends through the process - there was often initial sharing of personal events, vacations, family news, and other such matters which initially seemed off topic.

Although the conversations flowed freely, we generally had clear ideas about what we wanted to accomplish in a given phone call and meeting. If we had a pressing agenda - an upcoming class, or supporting the other in a presentation later in the week, we got right down to it. Jean, a consummate planner (says Sue), was usually the one to ask before the end of our allotted time for talking: "So what are we going to talk about next week?" If we didn't have enough time to finish planning, we communicated by e-mail.

We were also very free in bumping the Habits from the agenda if one of us wanted the other's feedback on a different project. We also used e-mail to organize ourselves, schedule our talks, reschedule our talks, trade drafts and exchange quick ideas during the week. While we didn't view e-mail as a medium for a lot of substantive virtual discussion, it was a very useful place for instant feedback or getting a thought or two down on paper and moving drafts back and forth. Others may find it to be a more reflective mode than we did.

These weekly conversations were supplemented in a number of ways. As a goal, we tried to get together in person once a month either at Sue's apartment in Manhattan or at Jean's house in North Haven, Connecticut. Typically the traveler would set aside the whole day and get an early start. Often without expectation the traveler was able to use

²⁰³ I think it is very important that we translate our law review articles and presentations into training materials for the bar and students. There is a void that we can fill and we have unique opportunities to collaborate with the bar to create greater access for clients.

the train time as reading or writing time and arrived at the host's house full of ideas and thoughts or maybe even moments of integration. Again the face-to-face time was loosely structured, allowing conversation in many avenues - sitting in front of a fire, taking a walk around the city or the neighborhood, cooking a meal and eating it. On a few occasions, Jean's student research assistants participated in some of the get-together. Again, the traveler often used the train ride back to write down some ideas or do some additional reading. (Both Sue and Jean found train time to be particularly fertile working time, almost justifying the trip in itself.)

In addition to using the insights that developed as we talked about our thinking and experiences and engaged in conversations with students and colleagues, we each spent time "teaching the other" about books, articles and materials we were reading. Sue spent her time reading scholarly and practical cross-cultural communication literature, materials for trainers and teachers who were developing cross-cultural materials and programs, and cross-cultural training [*102] materials for other professionals. Jean's specialty was to read off-point, finding parallels to what we were doing in her recreational reading. For several months, we discussed the parallels between the Habits and a book Jean had just read, The Inner Game of Tennis, ²⁰⁴ which focused on issues of mindfulness, non-judgmentalism and fact-based living. Jean also dubbed Habit Three "parallel universes" based on her avid interest in Star Trek and Star Wars. At another point, the experience of reading Marion Zimmer Bradley's Mists of Avalon ²⁰⁵ infused many conversations in which we talked about the ways in which stories could be viewed differently by different players. Jean also watched the Japanese classic, Rashomon, ²⁰⁶ and showed it to Sue, to explore the multiple stories/parallel universe concept. A chance encounter with a family member also exposed Jean to the Empathic Communicator, ²⁰⁷ which became the focus of our writing about conscious and unconscious competence and incompetence.

Sue used research assistants in a traditional way to locate cross-cultural material. Jean included her research assistants, Anjan Sahni, Cary Berkeley Kaye, and Leondra Kruger, in an ongoing conversation, which greatly enriched the process for the project and us. All three of the students had worked in the clinic, and two of them had been Jean's supervisees; in fact, throughout the process, our casework was an essential database for critical incidents for the Habits.

Typically, Jean would have both Sue and the research assistants on her schedule for one hour a week, so even if one was lost in client emergencies, the other would proceed. Jean kept her research assistants up to date on her conversations and work with Sue. They read and commented on drafts, road-tested the Habits, and suggested connections and ideas for research. Usually, one of the research assistants in each conversation would write up the conversation and e-mail it to all present, and to Sue. We often referred to our file of "minutes" later to look for trends and remember loose ends. On a number of occasions, Sue responded by e-mail with her reactions to the minutes and continued the conversation.

Later, as the project took shape, Jean structured the weekly meetings with Anjan, Cary and Leondra around refining each Habit, one at a time. Because Jean was on leave, the meetings took place at her house and could go for more than an hour. Sue planned two of her visits to North Haven to coincide with these meetings. Jean and [*103] Sue met in the morning; the five of us had lunch, and then discussed one of the Habits for two hours in the afternoon. There was something very rich about conversations that lasted for more than an hour. We were extremely lucky to have regular interlocutors who understood our lingo, lived with the ideas, edited our writing, tried them on in their daily lives, gave us frank feedback about them, advised us about the student perspective as we tried to plan teaching materials, and were extremely generous in giving us their time and ideas.

Sue's colleagues at CUNY agreed to teach the materials in cross-clinic classes that placed the students from each of the clinics into small groups. Sue used her sessions preparing her colleagues to use the materials as a starting

²⁰⁷ William Smiley Howell, Empathic Communicator (1982).

²⁰⁴ W. Timothy Gallwey, The Inner Game of Tennis (rev. ed. 1997).

²⁰⁵ Marion Zimmer Bradley, Mists of Avalon (1983).

²⁰⁶ Rashomon (1950) (directed by Akira Kurosawa; based on Ryunosuke Akutagawa et. al., Rashomon and Other Stories (reissue ed. 1999)).

point for the teaching section of the article. The last time the CUNY faculty taught the Habits, they videotaped the sessions for viewing and analysis. Sue consulted with Jean on planning these preparatory sessions and we both benefited from the feedback from faculty and students about how the methods worked.

Both the CUNY and Yale students were given quick writes at the end of each session asking them what worked and what they would change. We used a similar quick write at the AALS Clinical Teacher's Conference in Albuquerque and at a training session that Jean did for the Juvenile Rights Division of the New York Legal Aid Society. This feedback from colleagues and students encouraged us to pursue the project, gave us additional teaching ideas, and pointed out where we needed to elaborate our thinking.

B. Getting It on Paper

We have very different writing styles that complemented each other throughout this process. Sue is a careful note taker of reading, and note taker in discussions, and regularly reads back through her notes to find trends of ideas over time. Jean has a fascination for "one pagers": documents that can, on a single page, convey the large structure and some of the micro concepts of an idea. Jean was constantly fiddling with graphs, charts, and diagrams of the Habits throughout the process. At various points, when it came time to get something down on paper, Jean was often the person who did the sloppy first draft (either in the Internet chats described above, in a quickly dictated few pages, even once in a one pager prepared on a train ride on the way to a class taught at Sue's clinic at CUNY). Jean's writing style was more "stellar cartography," ²⁰⁸ in which she would use small periods of time (5 to 40 minutes) to advance a project by using dictation and collaborating with a research assistant.

[*104] At one point in the process, Jean used the Internet chat process with another friend to try to explain the Habits to an eager listener; the e-mail message turned into an early draft of the materials. Clearly technology offers many ways to have conversations that may be useful to the collaborative writing process.

Sue's writing took longer than Jean's, so she usually worked by adding to Jean's initial drafts, creating categories within the Habits. Because Sue was reading the cross-cultural literature, she found useful frameworks for describing the analysis that we were seeing in our work with students and clients. Each time we offered a class or gave a presentation, we added to the materials. By May 2000, we had a set of student materials for the Habits that we had used for several different classes. These materials were used for the demonstration at the 2000 AALS Clinical Teacher's Conference in Albuquerque, which by wonderful coincidence was all about culture.

Jean was able to use the conceptual work on the Habits and the student materials describing the Habits for a chapter on the Habits in her book. ²⁰⁹ Sue used her leave in 2000 and 2001 to do additional work on the materials and to write a more scholarly exposition of the Habits, which evolved into the article that appears in this issue of the Review.

II. Observations on the Process

No matter where we started and how far we got off track, our hour-long conversations invariably generated a number of insights about the Habits and our concerns about difference in our teaching and practice. Put in another way, the things that we shared with each other often had a kernel of something we had been talking about before, be it parallel universes, the camel's back analysis, or questions of differences and similarities.

We both comfortably move back and forth between the personal and the professional and view the personal as connected to our work. Thus, we never had "getting down to business" issues that can frustrate collaborators who have very different orientations to work and home.

In fact, it was fascinating and helpful to link the Habits to many parts of our lives. A regular pattern of going off on tangents or going off topic that happened naturally worked because we seemed to naturally find ourselves back on

²⁰⁸ Koh Peters, supra note 8 (Supp. 1998).

topic without any clear transitions being made. It also confirmed for us what we hoped was the daily usefulness of these concepts, since we were finding them applicable in many [*105] parts of our lives as we were trying to articulate them.

Our work styles and process values were similar enough that we worked well together. Both of us value collaborative work. Unlike people who get up from a meeting commenting about "getting back to work," as if the meeting was not work, both of us viewed the work we did together as "work."

We did not do identical amounts or kinds of work to advance the project. We had different opportunities at different times to put more or less work into the project and we accepted one another's contributions and limitations. We both learned from the on target/off target reading. Sue's openness to hear what many would have considered a tangent helped Jean make the connections that were actually there and contributed a great deal to both the fun and richness of our collaboration.

We both tended to use time rather than task as the main limitation on what we could accomplish when deadlines approached and so we got "as much done as we could in the time allotted" and were comfortable with an imperfect product. This allowed us to give classes and presentations before we had finished and to describe our work as a "work in progress."

Our process tended to be very story-centric. Very often our conversations would begin with a story from our recent week of work which turned out to have a natural connection to the Habits. In fact, throughout the process, our casework was an important database for critical incidents for the Habits. In conveying the Habits in our presentations and writing, stories have played a major part. We freely elicited stories from each other, dug for them, and had no concerns about making time for them because they tended to be very rich mines of critical incidents to examine.

In the course of our busy lives, spending fifty minutes a week on the phone together was very useful. These times were always oases of time when we had no agenda except to be in the moment together, trying to understand our project as well as possible. Jean often felt very refreshed after a conversation with Sue because it had been a fifty-minute period she had spent in the moment with someone else rather than planning for the future or debriefing the past. This process, which we grew to trust increasingly over time, kept the energy of the project alive as well. Staying in the moment, which of course is the core of many of the Habits, particularly Habit Four, turned out to be the most efficient as well as the most enjoyable working mode.

We really enjoyed teaching with each other and learned a number of lessons from these experiences. One of the benefits of this joint teaching was that we often saw different things in the classroom dynamics, [*106] yet shared a common database of experience and easily understood what the other had observed. One of us was an outsider, the other an insider to the class; that difference of perspective alone created different experiences for us. Interestingly, when we visited each other's schools, the home team teacher felt more anxiety about the class than did the visitor. In trying to think about the reason, we wonder whether knowledge of the students causes us to anticipate difficulties that a visitor could blissfully ignore? We wanted our students to embrace the Habits and the home teacher would see what the students had actually learned long after the blissful visitor departed? Or was it simply that we were inviting a friend to our home and we wanted her to like our family?

Finally, critical to the writing process was a spirit of playfulness and low risk of judgment by each other. Jean found the one-pagers easier to produce because she knew that Sue would feel totally comfortable trying out ideas in a class and then ditching them before the next presentation if they didn't work. Both of us knew that we could try stories or try out text in the writing and flag them in an e-mail to each other and get feedback on their appropriateness. Especially when time was of the essence in creating some written project for a conference presentation or a class, a spirit of forgiveness by the collaborator often helped things hit paper faster and easier.

III. A Parallel Process Between the Habits and the Process of Our Collaboration

The values of the Habits played a key role throughout the collaboration. Number one on the list was non-judgment. We each felt that our stories were heard and received gently and kindly even when, as in Habit Five analysis, they detailed stories of huge failures and even shame. The long-term process created a sense that any given encounter between us did not need to yield pay dirt because we would reap the benefits when we least expected it. The spirit of non-judgmentalism and playfulness in the process led naturally to a growing trust in which vulnerable and fragile matters could be shared more easily.

Similarly, looking at facts as opposed to judgments was a critical part of our process. For instance, in debriefing classes, inevitably one of us would feel concerned about a moment in class. Using Habit Five methodology, we were able to look at those less successful moments and move forward to determine how to avoid them in the future by looking at them carefully and dissecting them. It was a delight to be able to do that with each other, unafraid of judgment to come.

The mindfulness explored in Habit Four characterized our best discussions. Jean particularly remembers a phone conversation in [*107] which she was too distracted by a case emergency to pay proper attention and frequently had to ask Sue to repeat points and stories. It reminded Jean of one of our central points, which is that being present in the moment is not only more fulfilling but more efficient. Jean recognized that if she did not feel like she had time to have to have that conversation once, she certainly did not have the time to have it twice! Mindfulness throughout the process was also a wonderful counterpoint to other parts of our life which seemed very active and on the move at all times. Although we worked extremely hard by any standards during the meetings that we had, we looked forward to them as refreshingly quiet times in the midst of our busy lives.

Conclusion

Whether or not you develop a collaboration process that has features in common with ours, the most important piece of the process seems to be an ongoing trust in the process and in each other. There will be inevitable dry spells, periods like the beginning of the semester when not much work gets done on the project. Remember the principles of incubation and you may discover in the next conversation that things are all put together. The chart that first put down the habits was created on a train ride going to a class at CUNY and had followed a number of weeks of conversation that might have seemed somewhat diffuse to an outside observer.

For us, the combination of regular contact over a long period of time with interim deadlines but no final endpoint gave us a sense of abundance of time and resources which sped the process along. We wish for all of you who collaborate with others as rich a partnership as we've been fortunate to start, and urge you to create a process as idiosyncratic and eccentric as it needs to be to meet your needs and to make it fully enjoyable.

Copyright (c) 2001 Clinical Law Review, Inc. Clinical Law Review